Multinationals' Entry: Boon or Bane for Non-Frontier Economies?*

Guido Cozzi[†] Silvia Galli[†] Maria Bolboaca[†]

Abstract

Most countries in the world are not at the technological frontier, yet their economies grow and fluctuate. In this paper we set up a quantitative model of endogenous growth with business cycle fluctuations usable to analyze the medium frequency fluctuations in non-frontier countries. The growth mechanism is a Schumpeterian creative destruction framework, which is embedded into a real business cycle dynamic stochastic general equilibrium model, with standard and non-standard features. We allow for multinational firms entering the economy and challenging existing incumbents, and study the tension between their direct positive productivity contribution and their indirect negative contribution through the expected obsolescence of domestic innovators. We estimate the model using a full-information approach and show that multinationals' entry is both boon and bane for non-frontier economies.

JEL classification: E2, E3, F2, F4, O3, O4

Keywords: DSGE; Schumpeterian Growth; Medium Frequencies; Capital Inflows; Non-frontier Economies; Multinational Corporations.

^{*}The authors would like to thank seminar and conference participants at the University of Kent, University of St. Gallen, University of Lund, Spring Meeting of Young Economists (Brussels), and at the European Commission JRC (Ispra) for their insightful comments. We assume responsibility for all remaining errors.

[†]Institute of Economics, University of St. Gallen, Varnbüelstrasse 19, 9000 St. Gallen, Switzerland. E-mail: maria.bolboaca@gmail.com, guido.cozzi@unisg.ch, silvia.galli@unisg.ch.

1 Introduction

Non-frontier open economies, by their nature, are always exposed to Schumpeterian competition of foreign origins, i.e. firms operating in frontier economies, usually characterized by better knowledge of frontier technologies.¹ This is a point emphasized by a convincing empirical literature on the causes and effects of foreign direct investment (FDI) of multinational companies (MNCs) on the destination economies. In terms of triggers, a vast empirical literature (e.g. Calvo et al. (1996), Kim (2000), De Vita and Kyaw (2008), Forbes and Warnock (2012), Ahmed and Zlate (2014) and Ghosh et al. (2014), among others) shows that global factors, more than domestic developments, are to be accounted for the international capital flows to developing countries (DCs) or emerging markets (EMs). Regarding the positive effects of FDI for the destination country, Calvo et al. (1996) point to the economic growth enhancement and welfare improvement, achieved through increased consumption and investment.² However, the authors document the fact that large capital inflows have also many drawbacks. The procyclicality of foreign investments, in particular, has received great attention from scholars and policy-makers alike, because it may expose destination countries to abrupt reversals, capital inflows booms and busts being often linked to the business cycles in these countries.³

In this paper we address the question whether MNCs' entry is boon or bane for nonfrontier economies. The macroeconomics literature so far quarantines off one aspect of MNCs' entry at a time, business cycles effects being studied in isolation from those related to trend growth. However, the need to understand the consequences of MNCs' entry calls for a more integrated approach. To this end, we set up the first macroeconomic model able to account for the role played by MNCs' entry in driving economic fluctuations at high and low frequencies in a non-frontier small open economy (SOE).

We develop a dynamic stochastic general equilibrium (DSGE) model with endogenous adoption of an exogenously evolving technological frontier developed abroad. Domestic firms invest in research and development (R&D) in order to reduce the gap between their sector's productivity and the technological frontier, hoping to appropriate the associated profits by establishing intellectual property rights (IPRs) on it. They will be challenged by future domestic innovators, which leads to business dynamics typical of the Schumpeterian creative destruction literature stemming from Aghion and Howitt (1992). The mechanism of entry of better firms and exit of technologically obsolete firms emphasized by this literature (recently culminating in the works of Acemoglu et al., 2018) captures the importance of resource reallocation as an engine of growth that is instead missed in models of horizontal innovations, stemming from Romer (1990) and culminating with Comin and Gertler (2006), which unrealistically predict that firm exit is always bad for growth.

¹Broadly intended, inclusive not only of purely engineering blueprints, but also of better management practices, and whatever renders them more productive than local firms.

²Many empirical studies find supporting evidence in favor of the fact that FDI is growth-enhancing. For details, see Mello (1999), Alfaro et al. 2004; 2010, Adams and Opoku (2015), and Javorcik (2015), among others.

³See, e.g., Kaminsky et al. (2004), Reinhart and Reinhart (2009), Blanchard et al. (2010), Broner et al. (2013), Milesi-Ferretti and Tille (2014), Benigno et al. (2015), Araujo et al. (2017a), for empirical work on the linkages between capital inflows and the domestic business cycle.

While theoretical models without an explicit Schumpeterian creative destruction focus are biased in favor of positive effects of MNCs' entry, our creative destruction mechanism allows for more flexibility, as effects could overall be positive or negative, and depending on the data used in a calibrated or estimated version of our model, predictions may change. In fact, here we allow for a probability of entry of foreign MNCs transferring the foreign frontier productivity into the sector of the domestic economy in which they enter, but while this technological transfer from abroad is beneficial for the domestic economy's average productivity, it also comes at the expenses of the domestic firms in the sector, which are competed away by the entering MNC. Moreover, this reallocation effect is an additional source of creative destruction penalizing domestic innovators, which have undertaken the R&D investment targeting a partial catch-up with the world technological frontier, and then finding themselves overcome by a top quality foreign firm entering the sector. Hence, we are able to study also how the expectation of a probability of becoming obsolete by potential foreign entry affects the domestic innovator's R&D choice in the first place. While this has been studied in Schumpeterian growth models, such as Chu et al. (2014) and Cozzi and Impullitti (2016), it has never been cast in a medium frequency framework. Our DSGE model therefore has this novel feature of allowing foreign entry probability fluctuations which may result from foreign business cycles, but also from domestic and foreign policy changes - to be a source of fluctuations of the domestic economy.

Our model has a stronger foundation in economic theory than the purely econometric models that are predominant in the literature since, by allowing for inter-dependencies between different sectors, agents and markets in the economy, it can help shed light on whether and to what extend MNCs' entry and R&D are sources of economic fluctuations in non-frontier economies. We estimate the model following a full-information approach and using real data for Hungary to quantify the relative contribution of various shocks in explaining output growth in Hungary. The choice of Hungary is motivated by the fact that the central- and south-eastern European countries which are members of the European Union (EU), henceforth the CSEE,⁴ are a good example of non-frontier economies that have experienced strong economic growth and productivity gains accompanied by large FDI inflows but in the absence of considerable improvement in domestic innovation. Among them, Hungary is the country with the largest share of MNCs in the business sector in terms of value added, production, and turnover.⁵

Our estimation results, first interpreted through the impulse responses to various shocks introduced in this model, indicate that temporary shocks can cast a long shadow on the economy through both direct and indirect effects on the various model variables. In particular, stronger MNCs' entry, while discouraging domestic innovation, brings about an overall growth process in the country lasting far longer than the initial shock.

The estimation further suggests that, from a historical perspective, MNCs' entry shocks played an important role in driving the Hungarian GDP growth, especially before the Great Recession. After the accession to the EU of most CSEE countries, including Hungary, in 2004 the MNCs' entry shocks contributed the most to Hungary's GDP growth (up to 1

⁴The eleven CSEE countries which are members of the EU are Bulgaria, Croatia, Czech Republic, Estonia, Hungary, Latvia, Lithuania, Poland, Romania, Slovak Republic, and Slovenia.

⁵For details see Section 2.

percentage point in 2006Q1), acting as a catalyst of economic boom. However, from the onset of the financial crisis in 2007, we observe the drawback of the openness to foreign investment through the exposure to international shocks, MNCs' entry having an immediate negative impact on Hungarian GDP growth. According to the model estimation, negative MNCs' entry shocks depressed output growth by as much as 2.2 percentage points in 2009Q1, which amplified the economic downturn in Hungary.

An important result of this analysis is the procyclicality of capital inflows in Hungary, which is in line with the findings of various empirical studies. We perform a counterfactual exercise and find that, in the absence of MNCs' entry shocks, economic fluctuations in Hungary would have been of lower magnitude over the period 1995Q2:2019Q4. Our estimates indicate that MNCs' entry shocks not only make output more volatile but they have also dampened the recovery following both the global financial crisis and the sovereign debt crisis. The difference between the actual and the counterfactual series is most striking in the case of the financial crisis where we observe that, without the negative contribution of MNCs' entry shocks, output growth would have entered positive territory already in 2009Q2, while in reality this happened one year later.

Thus, we can conclude that MNC's entry proved to be both boon and bane for an EM such as Hungary since on the one hand it impacted positively economic growth, while on the other hand it was a significant source of business cycle amplification. On what concerns the government subsidies, the R&D subsidy shock had almost no effect on the observed variables, which highlights the insignificant role played by this shock in driving domestic productivity or output growth in both the short and medium run.

The paper is organized as follows. We first discuss the related literature. Section 2 then provides some motivating facts; Section 3 introduces the main model; Section 4 describes the calibration and estimation of the model, while Section 5 discusses the most important dynamic simulations; Section 6 concludes.

Related Literature

Our paper is related to several strands of literature. First, the paper develops on the novel literature stemming from Comin and Gertler (2006) in which productivity growth is endogenized in a DSGE framework. Successful models have recently been emerging, such as for example Roeger et al. (2008), Nuño (2011), Guerrón-Quintana and Jinnai (2014), Kung and Schmid (2015), Benigno and Fornaro (2017), Bianchi et al. (2018), Anzoategui et al. (2019), Bolboaca and Fischer (2019), Moran and Queralto (2018), and Cozzi et al. (2021). These papers mostly focus on advanced closed economies able to push forth the technological frontier. Some few exceptions are Comin et al.'s (2014) medium frequency two-country model, Roeger et al. (2016) and Cozzi et al. (2021), which allow some degree of technological inflow from abroad, Bolboaca and Fischer (2019), which studies a closed economy adapting an exogenously evolving technological frontier, Ates and Saffie (forthcoming), which extends the model of Klette and Kortum (2004) to a SOE with RBC features, Benigno et al. (2020), which studies the relationship between the surge of capital flows from DC to the United States (US) and the slowdown in global productivity growth, and Queralto (2020), which makes innovation subject to financial frictions in order to study slow recoveries after a financial crisis. However, none of them tackles the issue of a SOE exposed to MNCs' entry and unable to affect the world technological frontier, but undertaking R&D to adapt it to its local circumstances. Yet, most of the countries in the world are of this kind, including several European countries, and thus our model aims to fill this gap in the literature.

Another contribution to this literature is made with the introduction in our model of the preference specification of Schmitt-Grohé and Uribe (2012), which belongs to the class of intratemporal non-separable preferences developed by Jaimovich and Rebelo (2009). The Jaimovich-Rebelo preferences admit two polar cases, the utility function of King et al. (1988) with maximal degree of wealth effect and the preferences formulation with no wealth effect attributed to Greenwood et al. (1988). The estimation of the parameter governing the magnitude of the wealth elasticity of labor supply, based on the full-information approach and using Hungarian data, indicates that the utility function is practically defining the preferences of King et al. (1988). This implies that there is a strong income effect on labor supply, which is in line with the stylized facts for emerging economies (Aguiar and Gopinath, 2007).⁶

Our paper contributes also to the literature that quantifies the role played by capital flows in driving business cycles and growth. Much of the research in this literature aims to reproduce the stylized fact that fast growing, usually poorer, countries experience lower capital inflows, as showed in Gourinchas and Jeanne (2013), while slow growing, often richer, countries receive higher capital inflows. Many of these studies focus on the case of countries in Southern Europe and discuss why the surge in capital inflows in the early 2000s had a detrimental effect on resource allocation and productivity growth. Some recent examples are Reis (2013), Benigno and Fornaro (2012; 2014), Gopinath et al. (2017), and Jaccard and Smets (2020).

Another large body of this literature discusses how the financial integration of less financially-developed countries led to capital outflows from EMs, global imbalances, and a slowdown in frontier productivity (Caballero et al., 2008; Mendoza et al., 2009; Angeletos and Panousi, 2011; Prades and Rabitsch, 2012; Eggertsson et al., 2016; Sandri, 2014; Buera and Shin, 2017, among others).

Most of this theoretical literature abstracts from FDI flows, focusing instead either on aggregate capital flows or on their other two subcategories, namely portfolio investment or debt flows. There are some exceptions though, among which we may enumerate McGrattan and Prescott (2009) and McGrattan (2012), which extend standard growth models by introducing firm-specific technology capital that can be used in multiple countries, and Burstein and Monge-Naranjo (2009) and Ramondo (2014), which develop on the Lucas (1978) spanof-control model and allow for the movement of managerial know-how, interpreted as multinational production (MP). Several other papers introduce both trade and MP to quantify the overall gains from openness. See, e.g., Ramondo and Rodríguez-Clare (2013), which extend Eaton and Kortum's (2002) model to study how the various sources of complementarity and substitutability between trade and MP affect the gains from openness, and Irarrazabal et al. (2013), which builds on the model of Melitz (2003) and is in fact a quantitative application of Helpman et al. (2004) that focuses on understanding the frictions that rationalize export

⁶Queralto (2020) also employs a variant of the preference formulation of Greenwood et al. (1988) that has some similarities with the Jaimovich-Rebelo preferences. However, the author uses the method of minimum distance to estimate the parameter governing the wealth effect and employs data from the South Korean 1997 financial crisis for the estimation.

versus MP decisions. While we abstract from trade and a multi-country setup, our paper is complementary to these aforementioned studies in the literature modeling FDI flows as it allows to investigate the role of MNCs in bringing frontier technology and driving economic fluctuations in a non-frontier SOE, while at the same time to discuss the implications of MNCs presence for domestic R&D.

Another large literature to which this paper relates is the one that proposes quantitative frameworks to study economic fluctuations in EMs. Some prominent examples are: Neumeyer and Perri (2005), Bianchi (2011), Comin et al. (2014), Aguiar et al. (2016), Boileau and Normandin (2017), Houssa et al. (2019), Barrail (2020), Queralto (2020), Ates and Saffie (forthcoming). Among them, our model has some similarities with that of Houssa et al. (2019), which aims to explain business cycle synchronization between an EM, i.e. South Africa, and advanced economies (AEs), i.e. the US or G7, in that both SOE models are able to account for the influence of foreign shocks in the emerging economy and to demonstrate the important role played by these shocks in major crisis episodes. We differ in the categories of shocks considered as we focus on MNCs entry shocks, while Houssa et al. (2019) highlight the importance of commodity and credit supply shocks, and the fact that we introduce endogenous productivity growth in our setup to account not only for business cycle but also medium run fluctuations. With Queralto (2020) and Ates and Saffie (forthcoming) we share, in addition to the modeling approach, the purpose of connecting capital flows to productivity. However, while both models introduce financial frictions to study the dynamics following sudden stops in EMs, our aim is to link FDI with the economic fluctuations in a non-frontier economy. Comin et al. (2014) also introduce FDI but in a two-country model that explores the relative contribution of domestic and US shocks to Mexico's output fluctuations. Our model differs in many aspects but in particular in that we study the interplay between MNCs entry and domestic R&D in driving endogenous productivity in the non-frontier economy.

Finally, our paper is inspired by the empirical literature studying the impact of FDI flows into EMs, in particular those in the CSEE region. Iamsiraroj and Ulubaşoğlu (2015) review 108 empirical studies using data from around the globe and report 880 regression estimates of the effects of FDI on economic growth. The authors find that the distribution of these estimates is such that 43 percent are positive and statistically significant, 26 percent are positive and statistically insignificant, 17 percent are negative and statistically significant. and 14 percent are negative and statistically insignificant.⁷ They further investigate the relationship between FDI and output growth using data from a sample 140 countries over the period 1970 to 2009 and find that changes in FDI inflows do generate economic growth. A similar conclusion is drawn by Aizenman et al. (2013) who use cross-country data for 100 countries for the period 1990-2010 and find that the relationship between growth and lagged capital flows depends on the type of flows, economic structure, and global growth patterns. In particular they find a large and robust relationship between FDI and growth, but a weaker and less stable relationship between growth and equity flows. On what concerns the CSEE region, the literature provides also mixed results. Bačić et al. (2004) use aggregate data for the eleven CSEE countries over the period 1994 to 2002 and find that FDI cannot be accounted for the higher economic growth in the CSEE, while Fidrmuc and Martin (2011)

⁷See also Almfraji and Almsafir (2014) for a review of the literature on the relationship between FDI and economic growth.

employing data for the same group of countries between 1995 and 2009 conclude that exports and the stock of FDI in the CSEE region have a significant impact on economic growth. The findings of the latter are confirmed by Hlavacek and Bal-Domanska (2016), who use data for eight CSEE countries covering the period 2000-2012 and show that FDI positively affects economic growth in the CSEE countries, recording a stronger effect between 2009 and 2012.

The effect of FDI on productivity growth has been investigated to a lesser extent by the empirical literature. Kose et al. (2009) employ annual cross-country data over the period 1966 to 2005 for 67 AEs and DCs and find strong evidence that FDI and portfolio equity liabilities boost TFP growth.⁸ Bijsterbosch and Kolasa (2010) make an overview of the empirical evidence provided by the literature investigating this relationship in the CSEE countries and, using more recent industry-level data that covers nineteen sectors of eight of the CSEE countries (excluding Croatia, Bulgaria, and Romania) over the period 1995-2005, find that FDI inflows play an important role in driving productivity growth in the CSEE region.

In addition to the analysis of the potential benefits in terms of enhanced output and productivity growth, the empirical literature has also investigated whether capital inflows are an important driver of cyclical fluctuations in the destination countries. In particular, various empirical studies (e.g., Kaminsky et al. (2004), Broner et al. (2013), Lane (2015), Araujo et al. (2017; 2017), and references therein), based on data from a plethora of countries, have documented the fact that capital inflows tend to be procyclical and thus amplify the business cycle fluctuations in DCs and EMs. Our model contributes to this literature as it provides a unified growth and business cycle framework in which we can study the impact of MNCs entry on both the business cycle and the medium-to-long-term fluctuations in output.

2 Motivating Facts

The eleven central- and south-eastern European countries which are members of the EU, the CSEE, attracted about 3 percent of the total world FDI inflows between 1995 and 2019.⁹ While this may seem a small share when compared with the almost 40 percent of world FDI inflows that went to the whole Europe in the same period, the importance of FDI flows for the CSEE countries can be better assessed when measured as a share of GDP.¹⁰ Between 1995 and 2019, FDI inflows into the CSEE region averaged 4.15 percent of GDP. In contrast, the world average for the same period was 2.15 percent of GDP. Moreover, the average inflows in the CSEE was higher not only than in the EU15¹¹ (2.68 percent of GDP), but also

⁸See the references in Kose et al. (2009) for a review of the literature on the relationship between FDI and productivity growth using different levels of data aggregation.

⁹Poland received almost thirty percent of the inflows, being followed by the Czech Republic, Hungary, and Romania, the inward FDI flows to these countries representing slightly more than 70 percent of the group's total.

¹⁰Data source is the United Nations Conference on Trade and Development (UNCTAD).

¹¹The EU15 comprises the following fifteen countries: Austria, Belgium, Denmark, Finland, France, Germany, Greece, Ireland, Italy, Luxembourg, Netherlands, Portugal, Spain, Sweden, United Kingdom. These are the EU member countries that joined prior to May 2004. Throughout this paper, the EU15 is used as the representative group of countries for Western Europe and we disregard the withdrawal of the United Kingdom from the EU as this happened in January 2020, while our sample period ends in 2019.

than in both DCs and EMs (2.59 and 2.65 percent of GDP, respectively).¹² As illustrated in Figure 1, FDI inflows to the CSEE region represented a bigger share of GDP than in all the other groups of countries considered for comparison before the Great Recession, but the gap has narrowed afterwards.¹³ The biggest gap is observed in the period between the EU accession of most CSEE countries in 2004 and the trough of the global financial crisis in 2009. These large capital inflows helped fund a rapid growth in credit, consumption, and investment, and led to a widening of current account deficits and a buildup of external debt in the CSEE region. The Great Recession exposed the region's vulnerabilities, which reduced its competitiveness in attracting FDI and pushed the CSEE into a deep recession. Since the majority of foreign investors in the CSEE originate in Western Europe, the slowdown of FDI inflows accentuated once several eurozone countries entered the European sovereign debt crisis. The effect was stronger since not only European but also non-European investors feared further contagion in the EU and flocked to safer investments. However, for the past three years covered by our analysis we observe once more the pre-crisis trend, with the CSEE region attracting a larger share of FDI inflows as a percentage of GDP than the other groups of countries considered.



Figure 1: Average inward FDI flows as a percentage of GDP for the period 1995-2019.

The evolution of inward FDI flows and stocks in the CSEE over time is better reflected in Figure 2, where we observe the almost fourfold increase in FDI inflows from 15 billion US dollars in 1995 to 73 billion US dollars at their peak reached in 2007. Even compared to the level of inflows in 2019, the increase is twofold. In the same period, the inward FDI

¹²Note that data for FDI in the EU includes foreign investment of resident special purpose entities (SPEs), which are defined by Eurostat (2018) as enterprises formally registered with a national authority and subject to the fiscal and other legal obligations of the country in which they reside. This may render the stocks and flows of FDI in the EU15 overstated in relation to the economic impact of such investments (see Eurostat (2018) and Dellis et al. (2020) for more details). The presence of SPEs is particularly important for the data coming from Luxembourg, Netherlands, Cyprus, and Ireland. However, this is not the case of the CSEE countries in which the high share of FDI flows relative to GDP reflects the attractiveness of these countries to foreign investors.

¹³Note that the groups are those defined by UNCTAD.

stock of the CSEE group of countries increased from 33 billion US dollars to 827 billion US dollars. As pointed out by Kalotay (2017), what these numbers show is that the CSEE emerged as a new pole of attraction for global FDI within the enlarged EU. Furthermore, as illustrated in Figure 2, between 1995 and 2019 there were larger inward FDI flows than outward flows in the CSEE region, which indicates that the attractiveness of the CSEE region led to more capital flowing into the CSEE countries than out and thus made the region a net FDI recipient.



Figure 2: Inward FDI flows and stocks in the CSEE region for the period 1995-2019.



Figure 3: Comparison of GDP growth rates for the sample period 1995-2019.

There is a growing empirical literature that investigates the effects of international capital flows, in particular of FDI, on output and productivity growth in the CSEE. The reason is that these countries did not only experience large FDI inflows but also entered a process of catching-up with Western Europe after the collapse of the communist regimes in the region in the mid-1990s and their consequent transition to market economies (Bijsterbosch and Kolasa, 2010). In the years that followed the CSEE recorded significant economic growth and productivity gains. As it can be observed in Figure 3, real GDP of the CSEE grew at twice the speed of the EU15 GDP between 1995 and 2019.¹⁴ The higher average growth rate of the CSEE region indicates that there is an economic convergence towards the income levels of the most advanced Western European countries, even though the catching-up process has been slowed down by the global financial crisis.

One of the main contributors to the economic development of the CSEE region is considered to be productivity, which has experienced an acceleration in the same period (Georgiev et al., 2017). The average annual TFP growth was about 1.1% between 1995 and 2019 in the CSEE region, while it recorded an average of almost -0.1% in EU15 in the same period, which indicates that productivity in the CSEE countries grew faster than in the EU15.¹⁵ In Figure 4 it can be observed not only the difference between the productivity growth rates of the two main subgroups of the EU, but also the similarity to the evolution of output growth rates in these two regions which are displayed in Figure 3. Furthermore, there is empirical evidence which indicates that technological progress has been the main driver of the strong economic growth in the CSEE region before the Great Recession.¹⁶



Figure 4: Comparison of productivity growth rates for the sample period 1995-2019.

However, domestic innovation does not seem to be much accountable for this economic performance. The reason is that gross domestic expenditure on R&D in the CSEE region has been significantly lower than in the average European country between 1996 and 2018. Specifically, the CSEE countries invested on average less than one percent of GDP on R&D

¹⁴Data source is UNCTAD.

¹⁵Data source is the Total Economy Database (TED-2: Growth Accounting and Total Factor Productivity for 1990-2019, as of July 2020.

¹⁶See Levenko et al. (2017) for an analysis of the contributions to output growth of capital, employment, capacity utilization, and utilization-adjusted TFP growth before, during, and after the global financial crisis using CSEE data for the sample period 1996-2016.

projects in this period,¹⁷ while the average for the EU15 group of countries was about two percent of GDP. At the same time, the European and the world averages were close to 1.7 and 1.6 percent of GDP, respectively. The value of the R&D expenditure in the CSEE represented about 5.6 percent of the total expenditure made in the EU and 1.2 percent of the world total R&D expenditure.¹⁸ One may think that as the CSEE countries were and still are on average lagging behind those in the EU15 in terms of productivity levels, increased investment in domestic innovation in this region may have a higher marginal effect on productivity than in the rest of the EU. Nevertheless, when computing the cross-correlation between the growth rates of TFP and R&D expenditure, we find no evidence in favor of a lead-lag effect.¹⁹ In contrast, we find positive cross-correlations between inward FDI flows and different measures of productivity.²⁰ This contradicts Gourinchas and Jeanne (2013) who conclude that capital does not flow more to countries that grow faster. The situation of the countries in the CSEE region seems to be different than that of the DCs considered in the analysis of Gourinchas and Jeanne (2013) as they experienced higher productivity growth and at the same time attracted more capital between 1995 and 2019. While Gourinchas and Jeanne (2013) find that capital has been flowing "upstream" from DCs to the US, in the case of the CSEE we find that capital moved "downstream" to the poorer CSEE countries, investment coming mostly from the richer EU15 neighbors.

To conclude, the CSEE countries are a good example of non-frontier economies that have experienced strong economic growth and productivity gains accompanied by large FDI inflows but in the absence of considerable improvement in domestic innovation. While simple correlations of raw data point to the existence of a positive relationship between economic performance and FDI inflows, the empirical literature provides mixed results which depend on the econometric approach employed, sample of countries considered, data, and time span chosen. For this reason, to further investigate whether and to what extend MNCs' entry and R&D are sources of economic fluctuations in non-frontier economies, we employ a standard DSGE model for a SOE enriched with endogenous adoption of an exogenously evolving technological frontier developed abroad, which we describe in the next section.

Among the CSEE countries, we choose Hungary for our quantitative application of the model. Our choice was motivated by the fact that Hungary is the CSEE country with the largest share of MNCs in the business sector in terms of value added, production, and turnover.²¹ For example, in 2014 foreign penetration in terms of value added reached 53 percent in Hungary, more than 40 percent in the Czech Republic, Romania and Slovakia,

²¹We use data from Eurostat for foreign-owned companies (FATS dataset) and for all firms (SBS dataset) in the CSEE countries to investigate how important the MNCs are for the domestic economy and how they compare with local firms.

¹⁷Gross domestic expenditure on R&D (GERD) as a percentage of GDP is the total intramural expenditure on R&D expressed as a percentage of GDP. Data source is UNESCO Institute of Statistics (UIS).

 $^{^{18}}$ Total intramural R&D expenditure is measured in million purchasing power standards (PPS) at 2005 prices. Data source is UIS.

¹⁹We consider lags from 4 to 11 years and find no consistent pattern that could indicate that changes in lagged growth rates of R&D expenses lead to higher productivity growth.

²⁰We compute the cross-correlation between FDI inward flows as percentage of GDP and growth of TFP, growth of real labor productivity per person employed, or growth of real labor productivity per hour worked. The coefficients of correlation vary between 0.3 and 0.6 depending on the productivity measure employed. Data source for real labor productivity is Eurostat.

and less than 30 percent in Croatia, Poland and Slovenia. Moreover, as displayed in Table 6 in Appendix A, regardless of the indicator considered, the shares of MNCs in the Hungarian economy have been relatively stable over the period 2008-2018,²² which is in line with the assumption in Saffie et al. (2020) that Hungary reached a financially open steady state in 2008. Further details on the importance of MNCs in the CSEE region are provided in Appendix A.

3 The Model

In this section, we set up the model of our stylized non-frontier small open economy, which is cast in an infinite horizon in discrete time, t = 0, 1, 2, ..., with all agents having perfectly rational expectations. Growth is given by the evolution of the technological frontier A_t^{max} , which fluctuates exogenously around a deterministic trend rate $g_{\text{exo}} > 0.^{23}$

The economy is populated by households, goods producers, domestic innovators, multinational companies, and the government. Households, who are identical, consume the final good, invest in capital and R&D, and supply labor services to intermediate goods producing firms. Manufacturing is realized in two sectors. The final good producing sector is represented by competitive final goods producers, which combine the output of a continuum of intermediate goods producers. The intermediate goods producing sector operates under monopolistic competition and employs capital and labor services in its goods production. Domestic innovators use the R&D investment provided by households to produce a probability of filling the gap between their sectorial productivity and the technological frontier by targeting an intermediate technological level. A novelty of our model is that we allow foreign companies, in addition to domestic innovators, to enter the country and modernize the domestic industries. More specifically, we assume that, after domestic R&D takes place in a given sector, a multinational firm equipped with the frontier technology can enter the sector with an endogenous probability that depends on foreign direct investment. Lastly, the government's role is to partially fund the R&D expenditures of innovators and uses lump-sum taxes for this purpose. We further present their optimization problems in turn.

3.1 Manufacturing

A unique final good Y_t is produced at time t by perfectly competitive firms, which combine a continuum (normalized to a mass of 1 without loss of generality) of intermediate goods $i \in [0, 1]$ according to a constant returns to scale (CRS) production function

$$Y_t = \exp(\epsilon_t^{\text{temp}}) \left(\int_0^1 Y_{it}^{\frac{\xi-1}{\xi}} di \right)^{\frac{\xi}{\xi-1}}, \qquad (1)$$

²²Data source is the Hungarian Central Statistical Office.

 $^{^{23}}$ We discard population growth in this model because population has been shrinking in Hungary during the period considered in this paper, population decline being attributed not only to low birth rates but also to large out-migration. As migration can be reverted and the time series are not sufficiently long, it is not advisable to consider the trend observed in the past years as the steady state growth rate of population in this country.

with constant elasticity of substitution (CES) $\xi > 1$, and subject to a temporary disembodied productivity shock ϵ_t^{temp} .

It easily follows that the demand function for each intermediate good is

$$Y_{it} = Y_t \left(\frac{P_{it}}{P_t}\right)^{-\xi},\tag{2}$$

where P_t is the general price index

$$P_t = \left(\int_0^1 P_{it}^{1-\xi} di\right)^{\frac{1}{1-\xi}}.$$
(3)

Since we will focus on real variables, we normalize the final goods price index to 1.

The intermediate goods are produced under monopolistic competition. Each intermediate good i is produced using capital K_{it-1} and labor l_{it} according to a technology represented by the following Cobb-Douglas production function:

$$Y_{it} = A_{it} \left(l_{it} \right)^{\alpha} \left(\frac{K_{it-1}}{A_t} \right)^{1-\alpha}, \tag{4}$$

where α is labor elasticity, A_{it} is sector i's technological level, and $A_t = \int_0^1 A_{it} di$ denotes aggregate technology. As in Howitt and Aghion (1998), Nuño (2011), and Cozzi et al. (2021), with more sophisticated technologies, production becomes more capital-intensive. The intermediate good i's firm decides on the selling price, taking the demand of the final goods producer as given. Thus, its unconstrained profit maximizing price is

$$P_{it} = \frac{\xi}{\xi - 1} M C_{it},$$

with $\frac{\xi}{\xi-1}$ providing the mark-up over real marginal costs, MC_{it} . Moreover, the intermediate good *i*'s firm hires labor and capital, taking the real rental cost of capital, r_t^K , and W_t , the real wage, as given. Hence the factor demands of intermediate good i's producer are obtained by solving the following profit maximization problem

$$\max_{K_t, l_t} D_{it} = P_{it} Y_{it} - W_t l_{it} - r_t^K K_{it-1}.$$

Therefore, the demand equation for capital is

$$r_t^k = MC_{it}(1-\alpha)\frac{Y_{it}}{K_{it-1}},$$

and for labor is:

$$W_t = MC_{it} \alpha \frac{Y_{it}}{l_{it}},$$

In a symmetric equilibrium, intermediate goods producing firms hire capital and labor in the same ratio, which in turn equals the average ratio because they face the same factor prices. Therefore, they have the same real marginal cost, $MC_{it} = MC_t$. Going back to the pricing rule, having the same marginal cost, firms also charge the same price. From the demand specification, if all firms charge the same price, they must produce the same amount of output. Averaging sector *i*'s production function

$$Y_{it} = \frac{A_{it}}{A_t} \left(A_t l_{it} \right)^{\alpha} \left(K_{it-1} \right)^{1-\alpha}$$

and integrating over all sectors gives

$$Y_t = \exp(\epsilon_t^{\text{temp}}) \left(A_t l_t\right)^{\alpha} \left(K_{t-1}\right)^{1-\alpha}$$

which implies a standard growth accounting relationship

$$g_{Y_t} = \alpha g_{A_t} + (1 - \alpha) g_{K_{t-1}}.$$
 (5)

Since in the deterministic trend (balanced growth path) growth rates are constant, i.e. $g_{A_t} = g_A = g_{\text{exo}}$, and $g_K = g_Y$, from eq. (5) it follows that

$$g_K = g_Y = g_{\rm exo},$$

meaning that the trend growth rate of capital and GDP - and therefore that of investment and consumption - are all equal to the deterministic growth rate of frontier productivity. These considerations will be very useful when we recast our model in stationary form, which is required for numerical simulations.

Using these findings, we can rewrite the labor and capital demand equations in aggregate terms as it follows

$$r_t^k = \frac{\xi - 1}{\xi} (1 - \alpha) \frac{Y_t}{K_{t-1}},$$
$$W_t = \frac{\xi - 1}{\xi} \alpha \frac{Y_t}{l_t}.$$

3.2 Households

In this economy there is a unit continuum of identical households. The generic household $j \in [0, 1]$ maximizes its intertemporally additive utility function

$$U_{jt} = \mathbb{E}_t \sum_{s=0}^{\infty} \exp(\epsilon_{t+s}^{\text{beta}}) \beta^s u\left(C_{j,t+s}, l_{j,t+s}, S_{j,t+s},\right),$$

where the discount factor has a constant component $0 < \beta < 1$ and a normally distributed stochastic component driven by an exogenous shock ϵ_t^{beta} , which allows fluctuations in the time preference of households and thus affects both the marginal utility of consumption and the marginal disutility of labor.

The period utility $u(C_{jt}, l_{jt}, S_{jt})$ is a standard constant relative risk aversion (CRRA) utility function

$$u(C_{jt}, l_{jt}, S_{jt}) = \frac{\left(C_{jt} - hC_{jt-1} - \omega\theta^{-1}l_{jt}^{\theta}S_{jt}\right)^{1-\sigma} - 1}{1-\sigma},$$
(6)

which depends on household's consumption spending C_{jt} , labor supplied l_{jt} , and S_{jt} , defined as a geometric average of current and past habit-adjusted consumption levels according to:

$$S_{jt} = (C_{jt} - hC_{jt-1})^{\chi} S_{jt-1}^{1-\chi}.$$
(7)

This preference specification is similar to the one in Schmitt-Grohé and Uribe (2012), who introduce internal habit formation in consumption in the class of intratemporal non-separable preferences developed by Jaimovich and Rebelo (2009). The Jaimovich-Rebelo preferences allow the strength of the short-run wealth effects on labor supply to be controlled by $\chi \in [0, 1]$ and hence admit two polar cases, the utility function of King et al. (1988) with maximal degree of wealth effect ($\chi = 1, h = 0$), henceforth KPR, and the preferences formulation with no wealth effect ($\chi = 0, h = 0$) attributed to Greenwood et al. (1988), henceforth GHH.

As common in the literature, parameter $\sigma > 0$ measures the degree of relative risk aversion, $h \in [0, 1)$ governs the strength of habits in consumption, $\omega > 0$ is the disutility of labor, and $\theta > 1$ determines the Frisch elasticity of substitution of labor when $h = \chi = 0$.

The representative household in each period t receives labor incomes at wage W_t , profits D_{jt} from the $(1-F_t)$ firms they own, obtain interest $(1+r_{t-1})$ from foreign bonds B_{jt-1} , and rental rates r_t^K from their capital stock holdings K_{jt-1} . This income is spent on consumption C_{jt} , physical capital investment I_{jt} , the share $(1-Z_t)$ of R&D investment X_{jt} that is not refunded by the government in proportion to the R&D subsidy Z_t , lump-sum taxes T_{jt} , and foreign bonds according to the following budget constraint:

$$C_{jt} + I_{jt} + (1 - Z_t)X_{jt} + B_{jt} + T_{jt} = l_{jt}W_t + (1 - F_t)D_{jt} + (1 + r_{t-1})B_{jt-1} + r_t^K K_{jt-1}.$$

The household's law of motion of physical capital is

$$K_{jt} = \exp(\epsilon_t^{\text{MEI}}) I_{jt} \left[1 - \eta_I \left(\frac{I_{jt}}{I_{t-1}} \right) \right] + (1 - \delta) K_{jt-1},$$

where ϵ_t^{MEI} represents a marginal efficiency of investment (MEI) shock and $\eta_I\left(\frac{I_{jt}}{I_{t-1}}\right)$ denotes the quadratic adjustment cost to investment paid at time t and defined as

$$\eta_{I} = \frac{\gamma_{K}}{2} \left(\frac{I_{jt}}{I_{t-1}} - \exp\left(g_{I}\right) \right)^{2},$$

 $exp(g_I)$ being the growth rate of investment along the balanced growth path.²⁴

²⁴Notice that our timing convention for stocks, similar to that used in *Dynare (2011)* coding, assigns to stock variables the time date when they have been decided. For example, the investment in physical capital available for production in period t is decided at the end of period t - 1. Therefore, capital used in production at time t is indicated as K_{t-1} . Similarly for B_{t-1} , etc. This makes immediately clear that variables predetermined at time t have been accumulated in t - 1.

The household's first-order condition with respect to consumption is

$$\Lambda_{jt} = \exp(\epsilon_t^{\text{beta}})(C_{jt} - hC_{jt-1} - \omega\theta^{-1}l_{jt}^{\theta}S_{jt})^{-\sigma} - \beta h\mathbb{E}_t \exp(\epsilon_{t+1}^{\text{beta}})(C_{jt+1} - hC_{jt} - \omega\theta^{-1}l_{jt+1}^{\theta}S_{jt+1})^{-\sigma} - \Pi_{jt}\chi (C_{jt} - hC_{jt-1})^{\chi-1} S_{jt-1}^{1-\chi} + \beta h\chi \mathbb{E}_t \Pi_{jt+1} (C_{jt+1} - hC_{jt})^{\chi-1} S_{jt}^{1-\chi},$$

where Λ_{jt} and Π_{jt} are the Lagrange multiplier associated with the budget constraint and the law of motion of S_{jt} , respectively.

The household's first-order condition with respect to the risk free bonds is

$$\Lambda_{jt} = \beta \mathbb{E}_t \Lambda_{jt+1} (1+r_t).$$

The first order condition with respect to labor is

$$\exp(\epsilon_t^{\text{beta}})(C_{jt} - hC_{jt-1} - \omega\theta^{-1}l_{jt}^{\theta}S_{jt})^{-\sigma}\omega(l_{jt})^{\theta-1}S_{jt} = W_t\Lambda_{jt},$$

while for S_{jt} is given by

$$\Pi_{jt} = \exp(\epsilon_t^{\text{beta}}) (C_{jt} - hC_{jt-1} - \omega\theta^{-1} l_{jt}^{\theta} S_{jt})^{-\sigma} \omega\theta^{-1} l_{jt}^{\theta} + \beta(1-\chi) \mathbb{E}_t \Pi_{jt+1} (C_{jt+1} - hC_{jt})^{\chi} S_{jt}^{-\chi}.$$

In its optimal capital acquisition choice, each household compares the real price of a unit of physical capital, p_t^K , at the end of period t to the expected present discounted value of the next period rental rate of a unit of capital, r_{t+1}^K , plus the expected discounted value of the undepreciated part of capital remaining, $1 - \delta$, weighted by its new price p_{t+1}^K , according to

$$p_t^K \Lambda_{jt} = \mathbb{E}_t \beta \Lambda_{jt+1} \left[r_{t+1}^K + p_{t+1}^K (1-\delta) \right].$$

Then households optimally choose investment, by equalizing the cost of investment to the price of a unit of capital exclusive of adjustment costs

$$1 = \exp(\epsilon_t^{\text{MEI}}) p_t^K \left[1 - \frac{\gamma_K}{2} \left(\frac{I_{jt}}{I_{t-1}} - \exp\left(g_I\right) \right)^2 - \gamma_K \frac{I_{jt}}{I_{t-1}} \left(\frac{I_{jt}}{I_{t-1}} - \exp\left(g_I\right) \right) \right].$$

3.3 The R&D Sector

While this model economy characterizes a non-frontier country, its medium frequencies will be affected by the way endogenous domestic R&D adopts the evolving technological frontier developed abroad. More specifically, we will assume that the world technological frontier A_t^{max} evolves exogenously, with fluctuations around a constant trend growth rate g_{exo} , according to

$$A_t^{\max} = A_{t-1}^{\max} \exp(g_{\exp} + \epsilon_t^{\operatorname{Amax}}),$$

where ϵ_t^{Amax} is a shock process.

3.3.1 Domestic R&D

Domestic innovators use the R&D investment provided by households to produce a probability of filling the gap between their sectorial productivity A_{it-1} and the frontier A_t^{max} by targeting an intermediate technological level A_{it}^{target} measured by

$$A_{it}^{\text{target}} \equiv A_{it-1} + \phi_{t-1} \left(A_t^{\max} - A_{it-1} \right), \tag{8}$$

with ϕ_t defined as

$$\phi_t = \nu^{\left(1 - \frac{A_t}{A_t^{\max}}\right)},\tag{9}$$

and parameterized by $0 < \nu < 1$. Equation (9) allows the speed of convergence through domestic R&D to depend on the distance to the frontier of domestic average productivity, which is in line with the empirical findings indicating that countries closer to the frontier converge faster. Notice also in equation (8) that in case of success, the domestic innovator's technology will be dependent on the last period's technological level in the sector. This heterogeneity will reflect itself on marginal costs, prices, and production, implying that more advanced incumbents set the stage for higher productivity by domestic innovators in the sector. Each period there is a domestic entrepreneur able to attempt innovation. The success probability of domestic innovation depends on the entrepreneur's R&D expenditure X_{it} , expressed in final goods, according to the following innovation probability production function

$$n_{it} = 1 - \exp\left(-\frac{X_{it}\left[1 - \eta_X\left(\frac{X_{it}}{X_{t-1}}\right)\right]}{\varpi^{RD}A_{it}^{\text{target}}}\right).$$
(10)

For symmetry to physical capital investment, we assume that R&D investment is subject to a quadratic adjustment cost $\eta_X \left(\frac{X_{it}}{X_{t-1}}\right) = \frac{\gamma_{RD}}{2} \left(\frac{X_{it}}{X_{t-1}} - exp\left(g_X\right)\right)^2$. This captures in reduced form the idea that the past size of R&D laboratories sets a standard for the current research laboratory size, with researchers struggling when imposed abrupt changes.

Note also that, unlike Nuño (2011), we set the probability of innovation n_{it} to be always between 0 and 1 and increasing in R&D expenditure X_{it} , but with decreasing marginal product

$$\frac{\partial n_{it}}{\partial X_{it}} = \frac{1 - \eta_X \left(\frac{X_{it}}{X_{t-1}}\right) - \gamma_{RD} \frac{X_{it}}{X_{t-1}} \left(\frac{X_{it}}{X_{t-1}} - \exp\left(g_X\right)\right)}{\varpi^{RD} A_{it}^{\text{target}}} \exp\left(-\frac{X_{it} \left[1 - \eta_X \left(\frac{X_{it}}{X_{t-1}}\right)\right]}{\varpi^{RD} A_{it}^{\text{target}}}\right).$$

Our R&D production function (10) also features a degree of R&D difficulty that increases with the technological target, A_{it}^{target} , i.e. more sophisticated technologies are more difficult to achieve. This allows us to avoid the prediction of scale effects in adoption.

Moreover, unlike its linear counterparts used by Nuño (2011), Benigno and Fornaro (2017), and other important studies, the ability of this production function to always guarantee a non-negative and lower than 1 probability of innovation allows a lot of flexibility in the analysis of medium term frequencies in non-frontier economies, and sets a basis for several potentially useful theoretical and empirical exercises, including policy evaluations.

3.3.2 Multinational Firms' Entry

However, domestic innovators are not the only ones which can modernize the industries in the country. In fact, we assume that after domestic R&D is spent in sector i, a multinational firm equipped with the frontier technology A_t^{\max} can enter the sector with an endogenous probability f_{it} . Since both domestic and multinational firms operate under constant returns to scale, and given the assumption of price competition, the incumbent firm will exit the sector and the newly entering MNC will bring the frontier technology A_t^{\max} in that sector.

We remark that MNCs may also enter sectors already dominated by a MNC that entered in the past and hence bring the frontier productivity into the sector. Similarly, domestic innovators can challenge MNC incumbents endowed with past frontier technologies, now overcome by the current frontier. Also in this case, the domestic innovator can scale up their sector's productivity by targeting a better technology A_t^{target} determined by equation (8).

For the sake of symmetry, we model the probability of multinational entry f_{it} the same way as the success probability of domestic innovation. Thus, the MNCs' entry probability function depends on the investment M the foreign firm makes to enter the domestic market, i.e. FDI, according to

$$f_{it} = 1 - \exp\left(-\frac{M_{it}\left[1 - \eta_f\left(\frac{M_{it}}{M_{t-1}}\right)\right]\exp(\epsilon_t^f)}{\varpi^f A_t^{\max}}\right).$$
(11)

As in the case of physical capital and R&D investment, we assume that FDI is subject to the same type of quadratic adjustment cost $\eta_f \left(\frac{M_{it}}{M_{t-1}}\right) = \frac{\gamma_f}{2} \left(\frac{M_{it}}{M_{t-1}} - exp\left(g_M\right)\right)^2$. However, in contrast with the innovation success probability function, the probability of multinational entry is also subject to an exogenous shock ϵ_t^f that captures any stochastic fluctuations due to source or destination country factors that may influence the MNC entry decisions.

The foreign firm's expected profit maximization problem is given by:

$$\max_{M_{it}} f_{it} P_t^{S\max} - M_{it} R_{\mathrm{M}},$$

subject to equation (11), where $P_t^{S_{\text{max}}}$ is the nominal stock market value of the firm at the frontier technology and R_{M} is a constant MNC entry cost.

The first order condition is:

$$P_t^{S\max} \exp\left(-\frac{M_{it}\left[1-\eta_f\left(\frac{M_{it}}{M_{t-1}}\right)\right]\exp(\epsilon_t^f)}{\varpi^f A_t^{\max}}\right)$$
$$\frac{\left[1-\eta_f\left(\frac{M_{it}}{M_{t-1}}\right)-\gamma_f\frac{M_{it}}{M_{t-1}}\left(\frac{M_{it}}{M_{t-1}}-\exp\left(g_M\right)\right)\right]\exp(\epsilon_t^f)}{\varpi^f A_t^{\max}}=R_{\rm M}.$$

From our timing assumption that MNCs enter later in the period after the domestic innovator has attempted the innovation follows that the productivity in each sector i evolves according to the following intra-period lottery:

$$A_{it} = \begin{cases} A_t^{\max}, \text{ probability } f_{it-1}; \\ A_{it-1} + \phi_{t-1} \left(A_t^{\max} - A_{it-1} \right) \equiv A_{it}^{\text{target}}, \text{ probability } n_{it-1} \left(1 - f_{it-1} \right); \\ A_{it-1}, \text{ probability } \left(1 - n_{it-1} \right) \left(1 - f_{it-1} \right). \end{cases}$$

With these timing assumptions, the domestic R&D firm's expected profit maximization is expressed by:

$$\max_{X_{it}} n_{it} P_{it}^{\text{Starget}} \left(1 - f_{it} \right) - X_{it} \left(1 - Z_t \right) R_{\text{RD}} ,$$

subject to equation (10), where P_t^{Starget} is the nominal stock market value of the firm at the target technology and R_{RD} is a constant R&D cost.

The R&D investment first order condition is then:

$$P_{it}^{Starget}(1-f_{it}) \exp\left(-\frac{X_{it}\left[1-\eta_{X}\left(\frac{X_{it}}{X_{t-1}}\right)\right]}{\varpi^{RD}A_{it}^{target}}\right)$$

$$\frac{\left[1-\eta_{X}\left(\frac{X_{it}}{X_{t-1}}\right)-\gamma_{RD}\frac{X_{it}}{X_{t-1}}\left(\frac{X_{it}}{X_{t-1}}-exp\left(g_{X}\right)\right)\right]}{\varpi^{RD}A_{it}^{target}}$$

$$=\left(1-Z_{t}\right)R_{RD}.$$

Hence:

$$\frac{X_{it}\left(1-\eta_{X}\left(\frac{X_{it}}{X_{t-1}}\right)\right)}{\varpi^{RD}A_{it}^{\text{target}}} = \ln\left(\frac{P_{it}^{\text{Starget}}\left(1-f_{it}\right)\left[1-\eta_{X}\left(\frac{X_{it}}{X_{t-1}}\right)-\gamma_{RD}\frac{X_{it}}{X_{t-1}}\left(\frac{X_{it}}{X_{t-1}}-exp\left(g_{X}\right)\right)\right]}{\varpi^{RD}A_{it}^{\text{target}}\left(1-Z_{t}\right)R_{\text{RD}}}\right)$$
(12)

In equilibrium firm values are proportional to their productivity index, which implies that

$$\frac{P_{it}^{S\text{target}}}{A_{it}^{\text{target}}} = \frac{P_t^S}{A_t^{\text{target}}} = \frac{P_t^{Smax}}{A_t^{\text{max}}}, \text{ for all } i \in [0, 1],$$

where P_t^S denotes the value of an average intermediate good monopolist in this economy. Therefore equation (12) becomes

$$\frac{X_t \left(1 - \eta_X \left(\frac{X_{it}}{X_{t-1}}\right)\right)}{\varpi^{RD} A_t^{\text{target}}} = \ln\left(\frac{P_t^S \left(1 - f_{it}\right) \left[1 - \eta_X \left(\frac{X_{it}}{X_{t-1}}\right) - \gamma_{RD} \frac{X_{it}}{X_{t-1}} \left(\frac{X_{it}}{X_{t-1}} - exp\left(g_X\right)\right)\right]}{\varpi^{RD} A_t^{\text{target}} \left(1 - Z_t\right) R_{\text{RD}}}\right).$$

Notice from the term $1 - f_{it}$ that the risk of a MNC's vanifying the R&D investment of a domestic innovator will discourage domestic R&D investment.

Due to the two sources of creative destruction in our model economy, domestic and MNC innovations, the (average) stock market value of an incumbent intermediate good monopolist is expressed by the present discounted value of next period's dividends plus the continuation value of this firm scaled down by the lower than 1 probability of it not becoming obsolete due to domestic or foreign competition. Therefore its stock market valuation, after taking into account the household's stochastic discount factor, is:

$$\Lambda_t = E_t \left[\beta \Lambda_{t+1} \frac{D_{t+1} + P_{t+1}^S \left(1 - n_{it+1}\right) \left(1 - f_{it+1}\right)}{P_t^S} \right].$$

3.4 Sectorial Dynamics and Productivity Growth

Each sector will alternate the nationality of its leader based on the realization of the innovative entry processes. Therefore we cannot determine the nationality of the incumbent of a sector at a given time. However, since all innovative entry processes are assumed to be independent, we can appeal to the law of large number and use them to study the average frequency of sectorial incumbency changes, which is indeed deterministic. Remembering that multinationals can enter also sectors already dominated by a (potentially obsolete) multinational entered in the past, it follows that MNCs control a subset of the sectors of the economy of mass F_t , which evolves according to the following law of motion:

$$F_t = F_{t-1} + (1 - F_{t-1})f_{t-1} - F_{t-1}(1 - f_{t-1})n_{t-1}.$$

By aggregating the domestic and foreign innovations over the economy, and using the law of large numbers again, we can conclude that the average domestic technology evolves deterministically according to the following difference equation:

$$A_{t} = f_{t-1}A_{t}^{\max} + (1 - f_{t-1}) \left\{ n_{t-1} \left[A_{t-1} + \phi_{t-1} \left(A_{t}^{\max} - A_{t-1} \right) \right] + (1 - n_{t-1})A_{t-1} \right\}$$

3.5 Foreign Assets

This small open economy can free trade its homogeneous final good with the rest of the world at the price of 1. However, we do not restrict this country to always have a balanced trade, but rather allow it to accumulate net foreign assets or liabilities. Since there are various sources of fluctuations in this economy, the possibility of accumulating and decumulating net foreign assets allows its households to smooth consumption by borrowing in bad times and lending in good times. In this section we will then describe the asset and current account dynamics. Net foreign assets are expressed in terms of the household's net stock of foreign bonds, which in the aggregate evolves according to the following equation:

$$B_t = (1 + r_{t-1})B_{t-1} - D_t F_t + T B_t,$$

where TB_t denoted the domestic country's trade balance in period t. Trade balance is measured by the difference between the country's production of the final good and its aggregate expenditure in physical and R&D investment and consumption:

$$TB_t = Y_t - C_t - I_t - X_t.$$

Notice that we assume that households can, via a costless financial sector, also issue such bonds to finance their debt: hence, by consolidating all household's positions, country's net foreign liabilities will be represented by negative values of B_t . The real interest rate on foreign assets or liabilities is

$$r_t = \bar{r} + \psi \left[\exp \left(\frac{\bar{B}}{\bar{Y}} - \frac{B_t}{Y_t} \right) - 1 \right],$$

where \bar{r} is the steady state rest of the world interest rate, and $\psi > 0$ is the sensitivity parameter of the external debt elastic interest rate premium assumed here as in Schmitt-Grohé

and Uribe (2003). This allows us to study economic fluctuations of our small open economy using linear approximations in a relatively simple way, under the realistic assumption that if domestic assets are lower than a steady state ratio to GDP $\frac{\bar{B}}{\bar{Y}}$, or domestic liabilities are too high, then the country will have to pay an interest rate risk premium $\psi \left[\exp \left(\frac{\bar{B}}{\bar{Y}} - \frac{B_t}{Y_t} \right) - 1 \right]$.

The current account is obtained from the trade balance, by adding the incomés from foreign bonds and subtracting the dividends paid to the foreign owned firms (MNCs), that is:

$$CA_t = TB_t - D_tF_t + r_{t-1}B_{t-1}.$$

Notice that if the steady state foreign assets/liabilities are zero, the current account will be zero as well, while the trade balance will be positive and equal to the total dividends paid to the owners of the multinational firms.

3.6 Government

The government finances a part of the R&D investment performed by domestic innovators. The share of domestic R&D investment, Z_t , fluctuates stochastically according to shock ϵ_t^{RD} :

$$Z_t = Z \exp(\epsilon_t^{RD})$$

around a deterministic steady state value $Z \in (0, 1)$, with ϵ_t^{RD} following an AR(1) process.

The government raises revenue via lump sum taxes collected from households to finance its expenditures:

$$Z_t X_t = T_t,$$

3.7 Shocks

As previously mentioned, there are six exogenous sources of fluctuations in this economy: a shock to the frontier technology ϵ_t^g , a temporary productivity shock ϵ_t^{temp} , a shock to the household's intertemporal preferences ϵ_t^{beta} , a marginal efficiency to investment shock ϵ_t^{MEI} , a MNCs' entry rate shock ϵ_t^f , and a shock to R&D government investment subsidy, ϵ_t^{RD} , which follow an autoregressive process of order 1 and are described by the following equations:

$$\begin{split} \epsilon^{\text{temp}}_t &= \rho_{\text{temp}} \epsilon^{\text{temp}}_{t-1} + u^{\text{temp}}_t, \\ \epsilon^{\text{beta}}_t &= \rho_{\text{beta}} \epsilon^{\text{beta}}_{t-1} + u^{\text{beta}}_t, \\ \epsilon^{MEI}_t &= \rho_{MEI} \epsilon^{MEI}_{t-1} + u^{MEI}_t, \\ \epsilon^f_t &= \rho_f \epsilon^f_{t-1} + u^f_t, \\ \epsilon^f_t &= \rho_g \epsilon^g_{t-1} + u^g_t, \text{ and} \\ \epsilon^{RD}_t &= \rho_{RD} \epsilon^{RD}_{t-1} + u^{RD}_t. \end{split}$$

4 Estimation and Evaluation of the Model

In this section we analyze and simulate the model after setting numerical parameters. We solve the model numerically using a first order perturbation method. We use the solution

to the linear approximation of the detrended model around its deterministic steady state to find the equilibrium values of variables.²⁵ The parameters are set using both calibration and Bayesian estimation techniques.²⁶ We first discuss the calibration procedure and then continue with the estimation approach.

4.1 Calibration

We calibrate the steady state growth rate of the frontier technology to 0.9275/400 in order to match the roughly 0.9275 percent annual growth rate of TFP in the US over the period $1995Q2-2019Q4.^{27}$

We then fix the steady state gross real interest rate of the rest of the world to match an annual rate of 6 percent in the US, which we choose to be close to the mean after-tax return on business capital during the chosen sample period.²⁸ To satisfy the Euler equation, β equals 0.9875.

Parameter α is set equal to 0.56. We choose the calibration based on the average value of the share of labor compensation in Hungarian GDP at market prices over 1995-2019.²⁹ While the value for this parameter is smaller than the usual calibration of 2/3, it must be considered that this standard calibration in macroeconomic models is usually made for the US and using an average for a much longer timespan. Nevertheless, as discussed in Karabarbounis and Neiman (2013), the decline of the labor share has been registered in the majority of countries and industries since the 1980s, the main reason being a decrease in the relative price of investment goods that induced firms to shift away from labor and toward capital.

For the depreciation rate, δ , we set a value of 0.04/4, which implies an annual rate of depreciation of capital equal to 4 percent. This value of δ is the average depreciation rate of the capital stock for the period from 1995 to 2017 in Hungary.³⁰

We set the labor disutility parameter ω such that the steady state share of consumption in output is about 53%, which is the average share for the period 1995Q2-2019Q4 based on data from Eurostat.³¹

We choose for σ , the parameter that defines the curvature of the period utility, a value

²⁵See Appendix B for a description of the stationarized model.

 $^{^{26}}$ See An and Schorfheide (2007) for an overview of the literature on Bayesian estimation methods for DSGE.

²⁷We use the quarterly-TFP series for the US Business Sector, produced by John Fernald (version of March 05, 2020 and downloaded on November 12, 2020), as described in Fernald (2014)

²⁸As reported in Gomme et al. (2015), the return to capital was about 6 percent since 1995, it fell below 5 percent during the Great Recession, but rebounded afterwards to almost 7 percent in 2014.

²⁹Annual data is retrieved from the Total Economy Database (TED-2: Growth Accounting and Total Factor Productivity for 1990-2019, as of July 2020.) of The Conference Board. A similar result is obtained using data from AMECO for the compensation per employee as percentage of GDP at factor cost per person employed.

³⁰The average annual depreciation rate of the capital stock is retrieved from the Penn World Table, version 9.1 (Feenstra et al., 2015).

³¹We use GDP and main components (output, expenditure, and income) data from the dataset coded $[namq_10_gdp]$ of Eurostat.

of 2^{32} , which is a calibration often used in the macroeconomics literature.³³

Moreover, we calibrate $\theta = 2$, which implies that the Frisch labor supply elasticity is equal to 1 when $h = \chi = 0.^{34}$ We further calibrate steady state hours worked equal to 1/3, to match the fact that, according to the Labor Law in Hungary (Labour Code Art. 92 Sec 1), the standard working time is eight hours a day, which is equivalent to one third of the working day.

To calibrate the steady state value for the success probability of domestic innovation, n, we follow the approach of Comin and Gertler (2006) and consider the average time for the adoption of an intermediate good to be the inverse of this probability. For the US, it is standard to assume an adoption lag of about 10 years, given that between 1820 to 2003 in the US technologies needed about 19.8 years to be adopted (Comin and Hobijn, 2004; Comin et al., 2008; Comin and Hobijn, 2010; Comin and Mestieri, 2018). However, when considering the group of technologies that were invented after 1950 (i.e. cell phones, personal computers, internet usage, etc.), the average becomes 6.7 years (Eden and Nguyen, 2016). Since, for the case of Hungary we have data at most starting from 1995, and knowing that recent technologies are adopted faster than in the past, we assume an average adoption lag of 5 years in the US.³⁵ Using the sub-indices computed by Radosevic and Yoruk (2015) for technology capability in 42 countries, we obtain a scale of about 1/4 for Hungary.³⁶

³²We choose to calibrate this parameter instead of estimating it since the inclusion of this parameter in the list of estimated parameters makes χ , the parameter governing the magnitude of the wealth elasticity of labor supply, weakly identified. Since we do not have data for wages and employment to include as observables in the model and there is less econometric evidence on the value of χ , we prefer to estimate χ rather than σ . However, when applying the numerical optimizers to find the posterior mode of σ , based on a prior Gamma distribution centered at 4 and with a standard deviation of 1.5, we obtain a value of about 2.5, which is not significantly different from our calibration.

³³Most estimates of this parameter lie between 0 and 5, with 1, equivalent to a logarithmic utility, and 2, which indicates a higher degree of concavity, being the most common. Our calibration is motivated by Gandelman and Hernandez-Murillo (2015), who estimate the CRRA for 75 countries using data on self-reports of personal well-being from the Gallup World Poll. They find that estimates for both developed and DCs are concentrated in the vicinity of 1. However, the distribution of estimates for advanced countries contains relatively more mass of observations between 0 and 0.5, while the distribution for less advanced economies contains relatively more observations around 2. See Chetty (2006) and Barseghyan et al. (2018) for a survey of the literature on estimating risk preferences.

 $^{^{34}}$ Since estimates of the Frisch elasticity vary depending on the data and approach used for the estimation, with more recent estimates based on macro data indicating that the estimate is in the interval 1 to 2 (for details see Fiorito and Zanella, 2012, and Keane and Rogerson, 2012), or below 1 in the micro literature, we choose a value of 1 as a compromise between these findings (see Peterman, 2016, for a discussion of what explains the gap between the micro and macro estimates of the Frisch elasticity). Ríos-Rull et al. (2012) also indicate that results based on a Frisch elasticity of 0.72 or 1 are the most credible ones.

³⁵A shorter adoption lag in the US, e.g. 5 years, is also assumed in other recent papers such as Anzoategui et al. (2019).

³⁶We use the technology capability component of the first dimension, i.e. intensity of technology upgrading by types, of the composite technology upgrading index developed by Radosevic and Yoruk (2015) and Radosevic and Yoruk (2016). The intensity of technology upgrading by types dimension contains three components: production capability, technology capability and R&D capability. While there are no clear boundaries between the three components, some differences can be highlighted. For example, the production capability is more about producing efficiently and generating incremental innovations through improvements in products and processes. Technology capability differs in the sense that it performs advanced development

Using data from the Hungarian Central Statistical Office, we calibrate the steady state share of sectors dominated by multinationals equal to the average share of MNCs in the economy based on value added for the period 2008-2018.³⁷ Hence, the steady state value of variable F is 0.5. ³⁸ Plugging the steady state values for F and n in the equation defining the dynamics of the set of sectors dominated by MNCs, we obtain that the long-term probability of MNCs' entry, f is about 1.24 percent per quarter.³⁹

We employ data from Eurostat to set the steady state value of business expenditure on R&D (BERD) X as 0.6 percent of output.⁴⁰ and the R&D subsidy provided by the government to 11.5 percent of BERD.⁴¹ We further calibrate the R&D difficulty parameter and the R&D cost in order to match these moments.

We use data on FDI flows into Hungary, excluding special purpose enterprises (SPEs), and total GDP, both in million national currency, to set the steady state value of M equal to 4.15 percent of GDP, which is the average value for the period 1995-2019.⁴²

A summary of all parameter values is presented in Table 7 in Appendix C.5.

4.2 Estimation

The estimation of several of the structural parameters of the model is performed using Bayesian methods. We will first describe the observables, then present the prior distribution and posterior estimates, and finish the section with a summary of the estimation results.

4.2.1 Data

We perform the estimation using both quarterly and annual data. The reason is that, while most observables are at quarterly frequency, the series for domestic TFP growth and the share of BERD funded by the government are only available at annual frequency. Nevertheless, the implemented Kalman filter of *Dynare* deals with the missing information and thus allows for the model estimation with mixed frequency data.⁴³ Our dataset covers the period from

⁴⁰We use data from the dataset coded $[rd_e_gerdfund]$ of Eurostat and from STADAT tables 3.4.3 and 3.4.4 of the Hungarian Central Statistical Office. We divide the series for total R&D performed by the business sector by GDP (both in million national currency) and further average over the period 1995-2019.

by adapting prototypes for use in manufacturing, while R&D focuses on exploratory development which implies the invention of new prototypes (Radosevic and Yoruk, 2018).

 $^{^{37}}$ We use the data reported in the STADAT table 3.2.21.

 $^{^{38}}$ Using data from various OECD publications, e.g. OECD (2002), we constructed the series for 1995-2007 and using also this data the average share becomes 0.48. However, because the two values are relatively close and there were several revisions and methodological changes prior to 2008, we prefer to consider only the sample for the period 2008-2018.

³⁹With this calibration, the steady states for the success probability of domestic innovation, n, and the probability of MNCs' entry, f, are almost equal.

⁴¹We compute the share of R&D performed by the business sector that is funded by the government by dividing the series for the BERD financed by the government by the total BERD. We then report the average between 1995 and 2019.

⁴²We use FDI data from the Central Bank of Hungary, Table VIII. Balance of Payments, Foreign Direct Investment, International Investment Position (Foreign Direct Investments, according to BPM6 methodology).

⁴³Note that we have also performed the estimation using interpolated quarterly series derived from the annual series and the estimation results are similar.

1995Q2 to 2019Q4.

Our vector of observable variables is

$$\left[g_t^{\text{qobs}}, \Delta \ln Y_t^{\text{qobs}}, \Delta \ln C_t^{\text{qobs}}, \Delta \ln I_t^{\text{qobs}}, \Delta \ln A_t^{\text{aobs}}, Z_t^{\text{aobs}}\right],$$

where Δ denotes the temporal difference operator and the superscript indicates that the variable is observed at either quarterly *qobs* or annual frequency *aobs*. The time series of observed variables as introduced in the estimation are displayed in Figure 5.⁴⁴

The model is set at quarterly frequency. Thus, there is a frequency mismatch between the model and observed variable for domestic TFP growth and the share of BERD funded by the government. In this case, the observation equation describes the relation between the quarterly frequency model variable and the annual frequency observed variable. More specifically, we define the annual observed share of BERD funded by the government as the average of the model quarterly shares in the previous four quarters, i.e.

$$Z_t^{\text{aobs}} = 100 \left(Z_t + Z_{t-1} + Z_{t-2} + Z_{t-3} \right) / 4,$$

and the annual growth rate of domestic TFP equal to the change in stationarized quarterly average domestic productivity plus the trend growth over the respective year

$$\Delta \ln A_t^{\text{aods}} = 100\alpha \left(\ln a_t - \ln a_{t-4} + g_t + g_{t-1} + g_{t-2} + g_{t-3} \right).$$

The relationship between the quarterly model variables and observables is given by

$$g_t^{\text{qobs}} = 100g_t,$$

$$\Delta \ln Y_t^{\text{qobs}} = 100 \left(\Delta \ln y_t + g_t\right),$$

$$\Delta \ln C_t^{\text{qobs}} = 100 \left(\Delta \ln c_t + g_t\right),$$

$$\Delta \ln I_t^{\text{qobs}} = 100 \left(\Delta \ln i_t + g_t\right),$$

Note that $\ln j_t$, where $j = \{y, c, i, a\}$, is the natural logarithm of the respective stationarized variable as described in Appendix B.⁴⁵

⁴⁴Two outliers in the growth rate of consumption (2004Q1 and 2004Q2) and one in the growth rate of investment (2016Q1) were replaced with a three quarters centered moving average.

 $^{^{45}}$ We also estimated a version of the model with standard i.i.d. measurement errors on the observables, but the standard deviations of these shocks were weakly identified as indicated by various identification checks (see Ivashchenko and Mutschler (2020) for details on the identification diagnostics and tools implemented in *Dynare* and the issues arising from the lack of or weak parameter identification). Moreover, this finding was further confirmed by the fact that after the use of several different numerical optimizers to find the posterior mode, the curvature of the likelihood kernel for these parameters was flat, indicating that none of the measurement errors could be identified from the data. Nonetheless, the estimation of the other model structural shocks and parameters was robust to the inclusion of measurement errors. These results can be provided upon request.



Figure 5: Time series of observed variables as introduced in the estimation.

For the observable of the growth rate of frontier technology, we use the growth rate of TFP in the US, in percent change at a quarterly rate.⁴⁶ The series is produced by John Fernald,⁴⁷ as described in Fernald (2014).

All quarterly data for Hungary is from Eurostat. We construct real output by diving the nominal series for GDP at market prices by population and the implicit price deflator (2015=100). We proceed similarly to obtain the series for consumption and investment. We use the nominal series for household and NPISH final consumption expenditure, which we divide by population and the implicit price deflator to obtain real consumption. We define real investment as the nominal gross fixed capital formation divided by population and the implicit price deflator.⁴⁸ The quarterly log difference of these series are then introduced as observables in the model.

We use the annual growth rate of TFP in Hungary from the Total Economy Database⁴⁹ of The Conference Board as a measure of average domestic productivity growth.

As indicated in the previous section, we compute the share of R&D performed by the business sector that is funded by the government by dividing the series for the BERD financed by the government by total BERD.

4.2.2 Priors and Posteriors

The model parameters which we choose to estimate can be grouped into two categories. The first category comprises the parameters that characterize the law of motion of the exogenous

⁴⁶The original series is annualized, but we divide it by 4 to have it at a quarterly rate.

⁴⁷The series is reported in the dataset "Utilization-adjusted quarterly-TFP series for the US Business Sector" (version of March 05, 2020 and downloaded on November 12, 2020), available online at https://www.frbsf.org/economic-research/indicators-data/total-factor-productivity-tfp/.

⁴⁸By dividing each nominal series by its implicit deflator, we avoid the mismeasurement issues arising from the use of a common deflator as indicated by Fair (2020).

⁴⁹The data source is TED-2: Growth Accounting and Total Factor Productivity for 1990-2019, as of July 2020.

processes.⁵⁰ The second includes the parameters that are harder to be identified based on steady-state relationships among observable variables for Hungary.

As illustrated in Table 1, the priors for the estimated parameters are fairly diffuse and broadly in line with those chosen in the related literature. The prior distributions are displayed in Figure 18 in Appendix C.

To detail upon the choice of priors, for the consumption habit persistence we use a beta distribution centered at 0.7 and with a standard deviation of 0.1 since this parameter is bounded between 0 and 1. We choose the same prior distribution for χ , the parameter governing the magnitude of the wealth elasticity of labor supply, but with a mean of 0.5 and a standard deviation of 0.2 because we have less empirical evidence on the value of this parameter. For the priors of the parameters governing the degree of investment adjustment costs for either capital, R&D, or MNCs' entry, we assume a Gamma distribution centered at 2 and with a standard deviation of 1.5. This is the prior distribution imposed for the capital investment adjustment cost parameter in Smets and Wouters (2007), who were motivated by the findings of Christiano et al. (2005). Our approach only differs in that we choose a prior mean of 2 instead of 4, which is closer to the value chosen in the calibration of RBC models (see, e.g., Jaimovich and Rebelo, 2009). The prior mean for the substitution elasticity across goods is set equal to 5, implying that the gross markup in the goods production sector is centered at 1.25. Similarly to the distribution of the investment adjustment costs, we set the standard deviation equal to 1.5, which implies letting the data quite freely determine the size of these parameters.

For the sensitivity parameter of the interest rate premium ψ and the parameter governing the speed of productivity convergence ν , we use an Inverse Gamma prior distribution to ensure that the estimated parameters remain positive. We follow the approach of Adolfson et al. (2007) and set the prior mean of ψ equal to 0.01 with two degrees of freedom.

The priors for the parameters defining the exogenous processes are defined as in Smets and Wouters (2007). More specifically, the prior distribution for the autocorrelation of all structural shocks is centered at 0.5 and has a standard error of 0.2, while the standard deviation of the innovations is assumed to be Inverse Gamma distributed with a prior mean of 0.10 and a standard deviation equal to $2.^{51}$ By setting rather uninformative priors for these parameters, we let the data determine the importance of the disturbances. This is the usual approach in the estimation of small open economy models (see, e.g., Justiniano and Preston (2010) and Gelain and Kulikov (2011), among others).

The DSGE model is estimated based on observations from 1996:Q1 to 2019:Q4, conditioning on observations from 1995:Q2 to 1995:Q4. We first verify the identifiability of parameters by performing several local identification checks for the estimated parameters at the prior mean. The rank checks of Iskrev (2010), Komunjer and Ng (2011), and Qu and Tkachenko (2012), indicate that all parameters are identified.⁵²

⁵⁰Note that the covariance matrix of the innovations is diagonal.

 $^{^{51}}$ This implies that in terms of the parameters defining an inverse gamma distribution, the location parameter equals 0.0064 and the scale parameter is almost 2, which is equivalent to two degrees of freedom.

⁵²Even though all parameters are locally identifiable, the plots of the strength of identification and of the sensitivity component displayed in Figure 12 in Appendix C.1 indicate that parameter γ_{RD} might be weakly identified. This finding is confirmed in Figure 14 in Appendix C.1, which illustrates the parameter groups with the weakest identification. This seems to be due to high collinearity with respect to the effect

	Prior Distribution				Posterior Distribution					
Name	Density	Mean	S.D.	Mode	S.D.	Mean	Median	S.D.	5%	95%
Model Parameters										
h	Beta	0.7	0.1	0.855	0.021	0.855	0.856	0.022	0.811	0.894
χ	Beta	0.5	0.2	0.959	0.044	0.931	0.940	0.041	0.850	0.995
ξ	Gamma	5	1.5	7.048	1.083	7.047	6.964	1.155	4.890	9.338
γ_f	Gamma	2	1.5	0.008	0.017	0.025	0.019	0.021	0.000	0.065
γ_{RD}	Gamma	2	1.5	3.399	1.594	4.414	4.143	1.768	1.480	7.951
γ_K	Gamma	2	1.5	5.326	1.986	6.165	6.049	1.643	3.125	9.483
ψ	InvGamma	0.01	2	0.003	0.001	0.003	0.003	0.001	0.002	0.005
ν	InvGamma	0.1	2	0.044	0.063	0.104	0.065	0.167	0.021	0.255
Autocorrelation Coefficients										
ρ_g	Beta	0.5	0.2	0.249	0.118	0.275	0.269	0.112	0.064	0.490
$ ho_{\mathrm{temp}}$	Beta	0.5	0.2	0.994	0.007	0.989	0.990	0.007	0.975	0.999
$ ho_{ m beta}$	Beta	0.5	0.2	0.660	0.094	0.681	0.685	0.096	0.491	0.864
$ ho_{ m MEI}$	Beta	0.5	0.2	0.357	0.120	0.357	0.349	0.119	0.135	0.599
$ ho_{ m f}$	Beta	0.5	0.2	0.392	0.125	0.347	0.348	0.125	0.105	0.582
$ ho_{ m RD}$	Beta	0.5	0.2	0.835	0.072	0.798	0.810	0.080	0.639	0.936
Standard Deviations of Structural Shocks										
$SE_{u^{g}}$	InvGamma	0.1	2	0.010	0.001	0.010	0.010	0.001	0.009	0.012
$SE_{u^{\text{temp}}}$	InvGamma	0.1	2	0.014	0.001	0.014	0.014	0.001	0.011	0.016
$SE_{u^{\text{beta}}}$	InvGamma	0.1	2	0.070	0.011	0.078	0.077	0.011	0.058	0.100
$SE_{u^{\mathrm{MEI}}}$	InvGamma	0.1	2	0.173	0.062	0.201	0.197	0.050	0.110	0.304
$SE_{u^{\mathrm{f}}}$	InvGamma	0.1	2	0.207	0.095	0.318	0.291	0.118	0.149	0.549
$SE_{u^{\rm RD}}$	InvGamma	0.1	2	0.227	0.049	0.260	0.250	0.054	0.171	0.368

Table 1: Prior and Posterior Distribution of Estimated Parameters

We then follow Ivashchenko and Mutschler (2020) in their approach to find the posterior mode and the Hessian evaluated at the mode by sequentially looping over several optimization algorithms, taking the previously found mode as initial value for the next optimizer and starting from the CMA-ES algorithm.⁵³ We further use a Random-Walk Metropolis-Hastings sampling algorithm to obtain two Markov chains of 600,000 draws from the posterior distribution of the parameters and set the scale parameter such that the acceptance ratio is about one third for each chain.⁵⁴ A comparison of the prior and posterior densities is illustrated in Figure 20 in Appendix C.2.

Convergence of the two Markov chains is verified using the univariate and multivariate convergence diagnostics of Brooks and Gelman (1998). Figures 21 to 24 in Appendix C.3 indicate that the chains converge since in each plot the two lines stabilize horizontally and are close to each other. Furthermore, the trace plots illustrated in Figures 25 to 28 in Appendix C.4 are all stable, which confirms that the Metropolis-Hastings converges to a stable distribution.

We discard the first 300,000 draws in each chain as burn-in and sample uniformly 100,000 draws out of the remaining 300,000 to compute the posterior moments that are reported in Table 1. More specifically, Table 1 displays the prior distribution of all parameters, their posterior mode along with the approximate posterior standard deviation obtained from the inverse Hessian at the posterior mode, and lastly, the posterior mean and median along with the the 95% credible intervals. Since the mean and median of the posterior distributions are close for all parameters, we further discuss the results based on the posterior mean.

With the exception of the FDI adjustment costs, the posterior estimates for the parameters governing the role of real rigidities in the model are all far from zero, which indicates that real frictions are important for the model to fit the data. The estimated mean for the consumption habit formation parameter h equals 0.86, indicating that there is strong internal habit persistence in consumption. The value is at the higher end of the region in which most estimates for this parameter lie in the macroeconomic literature, but this is the usual outcome after the estimation of small open economy DSGE models.⁵⁵

Similarly, the mean estimates for the capital and R&D investment adjustment cost parameters are both higher than the prior mean. Among these parameters, the elasticity of capital investment adjustment costs has the highest mean, 6.2,⁵⁶ being followed by the elas-

of other parameters on the likelihood, in particular with γ_f and γ_K , as illustrated in Figures 16 and 17 in Appendix C.1. However, since the parameter is not unidentifiable, but only weakly identifiable, we attempt its estimation.

 $^{^{53}}$ The posterior mode for each estimated parameter is displayed in Figure 19 in Appendix C.2. This figure indicates whether a (local) mode was found, which is, ideally, when the estimated mode is placed at the maximum of the posterior likelihood.

 $^{^{54}\}mathrm{The}$ acceptance rate is 31.08% for the first chain and 30.51% for the second.

⁵⁵See Adolfson et al. (2008) for a comparison of the estimation results for closed and small open economy DSGE models and Havranek et al. (2017) for an extensive analysis of the factors that drive the diversity in the estimates of the habit parameter based on 81 published studies.

⁵⁶The value is higher than the posterior mean of 5.16 obtained by Jakab and Kónya (2016) for an SOE DSGE with nominal rigidities and search-and-matching frictions estimated using data for Hungary. However, it is smaller than the posterior mean of 6.96 obtained by Smets and Wouters (2003) in the estimation of a DSGE for the Euro area. Albonico et al. (2019) show than even when using the same model but different data, the estimates for parameters governing the capital investment adjustment costs differ across countries. Nevertheless, the credible interval for our posterior mean comprises most estimates in the related literature.

ticity for the R&D investment adjustment costs, 4.4, while the one for the FDI adjustment costs is estimated to be almost nil, 0.03. This shows the relatively low variability of capital investment in comparison to R&D investment and especially to inward FDI. The posterior mean of the adjustment costs parameter associated with the investment required for MNCs' entry being close to zero reflects the volatile nature of FDI.

The estimation drives the elasticity of substitution ξ to a larger value than the prior mean, which implies that the price markup of the monopolists is 1.165. While we could not find other estimates for Hungary and for the chosen time span, there is a large pool of estimates for various European countries or group of countries with which we can compare our result. However, the values vary considerably depending on the data and methodology employed. One of the most popular approaches in the recent literature is the cost-based method of De Loecker and Warzynski (2012),⁵⁷ which was used by De Loecker and Eeckhout (2017) and De Loecker and Eeckhout (2018) for computing weighted aggregate average price markups for the entire world economy, as well as for individual countries and regions. Using a sample that contains the financial statements of over 70,000 firms in 134 countries starting from 1980, De Loecker and Eeckhout (2018) find that the average global markup increased from close to 1.1 in 1980 to around 1.6 in 2016. While the authors do not report the average values for Hungary, in an illustration of the change in markup between 1980 and 2016, the country is located in the group that registered a decrease to no change in markup. This value is very close to our result, as the estimated posterior mean for the elasticity of substitution gives an average markup for Hungary that is close to the global and European average in 1980.

The posterior mean for the parameter ν is estimated to be equal to 0.1. This implies that the steady state probability of filling the domestic productivity gap by domestic innovators ϕ is 0.81, with a 95% credible interval of [0.68, 0.87]. Introducing these values in the steady state relationships, we get a steady state value of a, the domestic productivity variable, of 0.91, which can be interpreted as an indicator of the closeness to the technological frontier of Hungary.

The posterior mean of the parameter measuring the sensitivity of the interest rate premium with respect to the net foreign asset position is estimated to be significantly smaller than our prior, i.e. 0.01, which implies that there is a much lower sensitivity of the interest rate premium to the level of external debt than assumed a priori. The value is about 0.003, with a credible set of [0.002, 0.005]. This is larger than the small value calibration of 0.001 in Schmitt-Grohé and Uribe (2003) that is frequently used in the literature. However, the estimation of DSGE models tends to obtain larger values for this parameter, see e.g. Adolfson et al. (2007) and "Real Business Cycles in Emerging Countries?" (2010), while empirical work such as Kónya and Maduko (2020) indicates that the estimated elasticity parameter is highly dependent on how developed the chosen country is and the time span covered by the data used in the estimation.

The estimation of χ , the parameter governing the magnitude of the wealth elasticity of labor supply, gives a posterior mean of 0.93 with a credible set of [0.850, 0.995], which indicates that the utility function is practically defining KPR preferences. This implies that

 $^{^{57}}$ De Loecker and Warzynski (2012) adapted the method of Hall (1988) that was originally employed for the analysis of aggregate data to applications using micro data.

there is a strong income effect on labor supply, which Aguiar and Gopinath (2007) find to be in line with the stylized facts for transition economies.⁵⁸ There is a large array of estimates obtained in the literature that employ JR preferences, ranging from zero (Schmitt-Grohé and Uribe, 2012; Chang and Fernández, 2013) and thus supporting GHH preferences specification, through intermediate values indicating mild wealth effects (see e.g. Khan and Tsoukalas (2012) and Galí et al. (2012) for the result in the NK model without unemployment), to values close to one as in Siena (2021). However, to the best of our knowledge, there was no estimation of this parameter in a DSGE model with endogenous growth and hence we contribute to the literature with our finding.

The posterior means of the autocorrelation coefficients governing the stochastic processes of the structural shocks indicate that the most persistent disturbances are the domestic productivity shock and the shock to the share of BERD financed by the government. Both are in line in previous findings in the related literature. The estimate for ρ_{temp} is slightly above the standard value of 0.95 that is used for calibration in the RBC literature and usually obtained in DSGE model's estimations (see e.g. Smets and Wouters, 2003; 2007). The persistence coefficient for the shock to the share of BERD financed by the government is estimated to be 0.80. This is also a plausible result since government spending, in general, is often found to be a highly persistent process⁵⁹ and there is no reason to assume that the government spending directed to subsidize business R&D should behave differently. The other two domestic shocks, i.e. intertemporal preference and marginal efficiency of investment, have lower persistence coefficients. The relatively lower autocorrelation of these two shocks has emerged already in closed economy DSGE model estimations (Justiniano et al., 2011), but it is more prominent in small open economy setups (Gelain and Kulikov, 2011; Jakab and Kónya, 2016) since, as pointed out by Adolfson et al. (2007), the open economy aspects of the model are an additional source of frictions and disturbances. Nevertheless, the two shocks of foreign origin, the frontier technology growth shock and the MNCs' entry shock, are the least autocorrelated as they have the lowest estimated persistence coefficients.

Finally, the estimated standard deviations of shocks indicate that there are large differences in terms of volatility. The least volatile is the frontier technology growth shock, with a posterior mean standard error equal to 0.01. To the other extreme is the MNCs' entry shock that has the highest standard deviation (i.e. 0.32), which reflects the high volatility of inward FDI in Hungary.⁶⁰

We compute all remaining parameters and steady state values of the endogenous variables in stationarized form to satisfy the steady state equations and shock our model with the structural shocks.⁶¹ This allows to visually examine the dynamic effects of exogenous

 $^{^{58}\}mathrm{Aguiar}$ and Gopinath (2007) show that the correlation between output and hours is much lower in EMs than in developed markets.

⁵⁹This is the usual outcome of the Bayesian estimation of DSGE models, regardless of the data used and model setup. See, for example, Smets and Wouters (2003; 2007) Justiniano et al. (2010; 2011), or Gelain and Kulikov (2011), among others.

⁶⁰These large differences in terms of estimated standard deviations of shocks is not unusual in DSGE model's estimations. See, for example, Justiniano et al. (2011) for a closed economy DSGE in which the MEI shock is highly volatile, and Adolfson et al. (2007) or Justiniano and Preston (2010), among others, for similar outcomes in estimated open economy models, where shocks related to terms of trade and exchange rates prove to be relatively more volatile.

⁶¹See Table 7 in Appendix C.5 for the values of all parameters.

shocks in a non-frontier economy, from which medium term frequencies can be examined in a relatively intuitive way as it follows in the next section.

5 Estimation Results

5.1 Model Fit

Even though the main purpose of this model is not to best fit the data,⁶² we observe that it does a reasonable job in matching the actual volatilities of the observable variables. Table 2 reports the theoretical standard deviations of the observable variables generated by the model and their empirical counterparts. The model overestimates four volatilities and underestimates two of them, but half of the empirical standard deviations are contained in the 95% credible intervals generated by the model, while the other are marginally outside the bands.

		Data			
Variable	Mean	Median	5%	95%	
$\Delta \ln A^{aobs}$	1.881	1.841	1.404	2.271	2.042
$\Delta \ln Y^{qobs}$	1.378	1.370	1.207	1.545	0.945
$\Delta \ln C^{qobs}$	1.065	1.056	0.895	1.218	1.008
$\Delta \ln I^{qobs}$	4.018	3.900	3.075	4.963	2.913
g^{qobs}	1.065	1.053	0.883	1.237	0.602
Z^{aobs}	4.578	4.364	3.034	6.066	4.822

Table 2: Comparison of Standard Deviations

5.2 Variance Decomposition

As it can be observed in Table 3, the posterior mean variance decomposition of the observable variables to the various shocks of our model attributes a strong effect to the temporary productivity shock, which is usual in RBC models. The shock explains 70 percent of consumption growth, almost 50 percent of output growth and about 12 percent of investment growth. Among the other domestic shocks, the marginal efficiency of investment shock further explains about 78 percent of the variation in investment. It is followed by the intertemporal preference (discount factor) shock in a ranking of importance, the latter explaining about 23 percent of the variation in consumption and almost 19 percent of output's volatility.

An important finding is that the MNCs' entry shock has strongest effect on the observed measure of domestic productivity, explaining almost 97 percent of the variation in observed

⁶²The introduction of nominal rigidities or trade may improve the fit of the model. However, in that case the estimation would be done with data for a sample period starting from 2001 due to both data availability and change in monetary policy, i.e. adoption of inflation targeting in Hungary.

	u^g	u^f	u^{MEI}	u^{temp}	u^{beta}	u^{RD}
$\Delta \ln A^{aobs}$	2.98	96.54	0.02	0.44	0.01	0.02
	[0.89; 5.79]	[93.53; 98.79]	[0.00; 0.04]	[0.01; 0.82]	[0.00; 0.03]	[0.00; 0.06]
$\Delta \ln Y^{qobs}$	1.10	28.33	2.63	49.21	18.65	0.09
	[0.41; 2.08]	[17.41; 40.01]	[1.08; 4.45]	[37.35; 61.11]	[9.69; 28.05]	[0.00; 0.26]
$\Delta \ln C^{qobs}$	2.03	4.32	0.34	70.10	23.18	0.04
	[0.71; 3.84]	[0.46; 9.60]	[0.07; 0.74]	[55.56; 83.01]	[12.19; 35.55]	[0.00; 0.11]
$\Delta \ln I^{qobs}$	0.15	9.72	77.50	12.16	0.37	0.10
	[0.05; 0.30]	[3.13; 18.09]	[65.57; 88.46]	[5.59; 19.46]	[0.03; 0.99]	[0.00; 0.31]
g^{qobs}	100.00	0.00	0.00	0.00	0.00	0.00
	[100; 100]	[0.00; 0.00]	[0.00; 0.00]	[0.00; 0.00]	[0.00; 0.00]	[0.00; 0.00]
Z^{aobs}	0.00	0.00	0.00	0.00	0.00	100.00
	[0.00; 0.00]	[0.00; 0.00]	[0.00; 0.00]	[0.00; 0.00]	[0.00; 0.00]	[100; 100]

Table 3: Posterior mean variance decomposition (in percent)

TFP. This shows how important FDI inflows are for domestic productivity and consequently for output of whose variation the MNCs' entry shock explains about 28 percent. Note that since MNCs' entry is endogenous, domestic shocks may also influence the decision of foreign companies to enter this country. However, the variance decomposition based on observed data indicates that over the period considered domestic factors did not play an important role in MNCs' entry decision given how little domestic shocks explain of TFP's volatility. This result is in line with various empirical studies that attribute the leading role in driving capital flow volatility to global factors (see, e.g., Kim (2000), Forbes and Warnock (2012), and references therein).

On what concerns the governmental subsidies, the R&D subsidy shock has almost no effect on the observed variables, which highlights the insignificant role played by this shock in driving domestic productivity or output growth.

5.3 Impulse Responses

We describe the general equilibrium effects of each shock in isolation, by studying the impulse response function (IRF) plots with the response of the endogenous variables, expressed in deviation from the steady state value, to a standard deviation change of the white noise disturbance for a horizon of 30 years (120 quarters).

One important observation from analyzing the impulse response functions obtained from our model is that temporary shocks can cast a long shadow on the economy. An interesting and very powerful shock, which is at the core of the contribution of this paper, is the positive shock ϵ_t^f to FDI flow into the country. As a response to this temporary shock, the set of sectors dominated by MNCs increases for a long time, as illustrated in Figure 6.



Figure 6: Impulse responses to a one estimated standard deviation MNCs probability of entry shock. The unit of the horizontal axis is quarters and of the vertical axis is deviations from steady state values.



Figure 7: Impulse responses to a one estimated standard deviation MNCs probability of entry shock. The unit of the horizontal axis is quarters and of the vertical axis is deviations from steady state values.

The resulting increased flow of frontier technology into the domestic economy is reflected in the positive medium term effects on productivity, output, consumption, capital investment, and labor, with resulting dynamics displayed in Figures 6 and 7. Notice that this happens despite the drop in domestic R&D, discouraged by the increased probability of creative destruction for domestic innovators. This is also reflected in the initial drop of the stock market value of domestic firms. Hence, stronger MNCs' entry, while discouraging domestic innovation, brings about an overall growth process in the country lasting far longer than the initial shock. It is important to note that physical capital accumulation amplifies the medium-run effects on output and on workers wages, whose marginal productivity increases.

On the downside, we observe that MNCs' entry, by increasing the share of sectors dominated by multinational firms, implies that more dividends are repatriated by the owners of the foreign affiliates to their home country. This acts as a wealth shock that leads to a decline in the country's net foreign asset position. Households are poorer since they receive less dividends given that some of the companies they own became obsolete and thus need to adjust their consumption. This negative effect on consumption is the reason why the overall positive effect that we observe in the impulse response of consumption to the MNCs' entry shock is much smaller than in the one of output, both in the short and medium run. Since domestic demand for goods does not increase much, more output is thus available for trade, this leading to the improvement in the trade balance. The positive effect of the MNCs' entry shock on the trade balance is even stronger in the medium run when physical capital investment slows down.

The investment boom, the increase in dividend payments of MNCs, along with the decline in savings leads also to a a widening of the current account deficit. This confirms the findings of Calvo et al. (1996), who show that in the 1990s the largest recipients of capital inflows in Asia and Latin America also experienced larger current account deficits. Furthermore, using data for 58 DCs over the period 1978-1995, Bosworth et al. (1999) find that capital flows to DCs were in fact used to finance current account deficits and most of this resource transfer was directed into domestic investment and not consumption, which is in line with our results.



Figure 8: Impulse responses to a one estimated standard deviation R&D subsidy shock. The unit of the horizontal axis is quarters and of the vertical axis is deviations from steady state values.

Another valuable exercise is the analysis of the medium run response of the economy to an R&D subsidy shock. By introducing this shock in the model, we let the government temporary raise the share of the BERD investment that it finances and hence to tax more the representative household for this period. As displayed in Figures 8 and 9, the overall R&D investment increases, which leads to a higher probability of innovation. Unfortunately, this does not drive an increase in domestic productivity since MNCs' entry drops, discouraged by the policy action, and domestic R&D cannot compensate for the loss of their contribution to productivity growth. Therefore, through this channel in fact an R&D subsidy shock leads to a reduction instead of an increase in average productivity. Temporarily higher taxes and the decrease in the marginal product of capital leads to a drop in physical capital investment, which amplifies the negative effect of the decrease in endogenous productivity on output. However, households have more income coming from dividends payment since there is less foreign competition on the domestic market. This translates into an increase in consumption both in the short and medium run. Once the effect of the R&D subsidy shock fades away and the lump-sum tax on households returns to its previous level, savings increase and this is reflected in the medium run increase in net foreign assets. Nevertheless, all impulse responses indicate much smaller effects of the R&D subsidy shock in comparison with those of the MNCs' entry shock.



Figure 9: Impulse responses to a one estimated standard deviation R&D subsidy shock. The unit of the horizontal axis is quarters and of the vertical axis is deviations from steady state values.

5.4 Historical Shock Decomposition of Real Output Growth

In this section we quantify the contribution of exogenous shocks in explaining observed real GDP growth through the lens of the estimated model. In Figure 10 the continuous black line represents the observed quarterly real GDP growth rate over the period 1995:Q2 -2019Q4. The vertical bars indicate the contribution of the respective exogenous shocks to explaining the data in each period. Bars above the horizontal axis represent positive shock contributions, while bars below the axis indicate negative contributions.

Figure 10 illustrates the important role played by MNCs' entry shocks in driving the Hungarian GDP growth, especially before the Great Recession. The contributions of these shocks, marked as the light green bars, are in line with the periods of FDI policy identified by Antalóczy et al. (2011). This is an extremely valuable result given that we do not include data on FDI in our model estimation. According to Antalóczy et al. (2011), the first period started in 1988 and lasted until the mid 1990s, the second period lasted until the end of 2003, and the third period ended in the Great Recession of 2007-2009. We further consider that the fourth period began with the second premiership of Viktor Orbán in 2010 and was still ongoing at the end of 2019 when our sample period ended.


Figure 10: Historical shock decomposition of observed output growth. The continuous line represents the observed time series. Stacked vertical bars indicate the estimated contribution of different shocks in a given period. The contribution of all shocks recovers the observed variable.

While we do not have data for the first period, between 1988 and 1995, Hungary was not only one of the main recipients of foreign investment but also one step ahead the other CSEE countries in attracting privatization-related FDI for several reasons. First of all, the country had been experimenting with marketization since 1968 and was closer to a market economy than the other neighboring economies that started the transition in 1989. Moreover, the Hungarian government passed various laws that facilitated the selling of state enterprises to foreign investors earlier than the other countries in the region.⁶³

By mid 1990s, the share of privatization-related FDI became negligible in Hungary since most state-owned companies had been already privatized, while the privatization process was taking off in the other Visegrád countries (Sass, 2004). This explains why at the beginning of the second period, between mid 1990s and 2000, which is also covered by our data, we observe in Figure 10 that the effect of the foreign firms' entry shock on Hungary's GDP growth is relatively small. However, according to Antalóczy et al. (2011), being a forerunner in terms of attracting privatization-related FDI gave a good signal to other potential foreign investors. At the same time, the more stable economic and political environment obtained in this second period in Hungary, along with various changes in the FDI policy, which became more transparent and normative, made Hungary a valuable destination for further FDI inflows. Hence, despite the regional competition for FDI coming from neighboring countries such as the Czech Republic or Slovakia, which introduced generous systems of incentives for foreign investors, Hungary continued to attract significant FDI inflows. In the period between 2000 and 2003, these inflows account for more than half of the positive contributions to Hungarian GDP growth, according to our estimates.

The third period started in 2004 due to the accession to the EU of most CSEE countries, including Hungary. In this period, the inflow of FDI in Hungary boosted because of the free movement of goods, services, capital, and labor implied by the EU Single Market. Moreover,

⁶³See, e.g., Klaudt (1995), Antalóczy et al. (2011), and references therein, for more information on the post-revolutionary economic laws passed in Hungary and their implications for foreign investment.

the EU integration increased the confidence of foreign investors in the future economic development of the region and its political stability (Becker et al., 2010). Furthermore, having many of the countries in the region joining the EU at the same time implied a reduction in the regional incentive competition as they all had to change their FDI policies along the lines of the EU (Antalóczy et al., 2011). This is reflected in Figure 10 where we observe that after 2004 the MNCs' entry shocks contribute the most to Hungary's GDP growth (up to 1 percentage point in 2006Q1), acting as a catalyst of economic boom.

However, from the onset of the financial crisis in 2007, we observe the drawback of the openness to foreign investment through the exposure to international shocks, MNCs' entry having an immediate negative impact on Hungarian GDP growth. This may be attributed to the fact that the US is the largest non-EU investor in Hungary (U.S. Department of State, 2019). The negative effect of these foreign shocks has been the largest in 2008-2009 as the financial crisis spread globally and sparked the Great Recession, leading to a reduction in global foreign investment. Moreover, with several eurozone countries entering the European sovereign debt crisis by the end of 2009, investors feared further contagion of other European countries and flocked to safer investments. Our estimates indicate that negative MNCs' entry shocks depressed GDP growth by as much as 2.2 percentage points in 2009Q1, which amplified the economic downturn in Hungary.

In 2010 the parliamentary elections in Hungary led to the beginning of the second premiership of Viktor Orbán and a change in political attitude towards foreign presence. Therefore, we consider this to be the start of the fourth period in terms of FDI policies, which lasted at least until the end of our sample period, i.e. 2019Q4. In this period, FDI inflows increased in the CSEE region, but did not reach the pre-crisis levels. In Hungary, the shift from a democratic towards an authoritarian populist regime, increased corruption, the introduction of unpredictable industry-specific taxes and changes in regulatory policies favoring government-controlled companies, strong anti-migrant rhetoric and actions, have reduced foreign presence in various sectors and further discouraged MNCs' entry in general (Bershidsky, 2017; Adám, 2019; U.S. Department of State, 2019; Eder, 2021). This explains the smaller contribution of MNCs' entry shocks after the Great Recession. However, the regulatory burdens aimed at decreasing foreign market share did not target foreign industrial investors, which in fact have continued to be lured in Hungary with special tax breaks and direct subsidies for large investments in the manufacturing sector, especially after 2016 (Benner and Reinicke, 2017; Sarnyai, 2018; U.S. Department of State, 2019; Eder, 2021). This coincides with the period in which we observe a persistent positive contribution of MNCs³ entry shocks on Hungary's GDP growth, according to our estimates (see Figure 10). An important result of this analysis is the procyclicality of capital inflows in Hungary, which is in line with the findings of various empirical studies.⁶⁴ This is apparent in Figure 10, where positive contributions of MNCs' entry shocks are observed in boom periods, while negative contributions are mostly present in periods of recession.⁶⁵ Araujo et al. (2017a) point to the

⁶⁴See, e.g., Kaminsky et al. (2004), Broner et al. (2013), Araujo et al. (2017a), and references therein, for empirical evidence in favor of the procyclicality of aggregate and disaggregate capital inflows based on data from a plethora of countries.

⁶⁵Having foreign sources of fluctuations driving capital flows is supported by previous empirical research. For example, De Vita and Kyaw (2008) show that shocks to foreign output are among the most important forces explaining the variations in capital flows to DCs.



Figure 11: Comparison of observed output growth and the counterfactual obtained from the contribution of all shocks except MNCs' entry. The black line represents observed output growth. The green crossed line defines the counterfactual obtained from the estimated contribution of all shocks except MNCs' entry to the series.

dangers of procyclical capital inflows, in particular to the fact that they exacerbate macroeconomic fluctuations as well as amplify the domestic financial cycle. The empirical analysis of Kaminsky et al. (2004) further highlights that in DCs - and even more so in EMs - periods of capital inflows are associated with expansionary macroeconomic policies and periods of capital outflows with contractionary macroeconomic policies. Thus, macroeconomic policies in these countries mostly reinforce the business cycle, which is in stark contrast with the appropriate countercyclical measures that should be taken in order to stabilize the cycle. Valentinyi (2012) states that this has also been the case of Hungary, where the expansive fiscal policy and the build-up of a large external debt prior to the Great Recession made Hungary one of the most financially vulnerable countries in the CSEE. This is borne out in Figure 11, where we compare observed output growth and the counterfactual obtained from the contribution of all shocks except MNCs' entry.

Figure 11 shows that, in the absence of MNCs' entry shocks, economic fluctuations in Hungary would have been of lower magnitude over the period 1995Q2:2019Q4, with a coefficient of variation of 1.95 for the counterfactual series compared with a coefficient of 2.07 for the observed series. Our estimates indicate that MNCs' entry shocks not only make output more volatile but they have also dampened the recovery following both the global financial crisis and the sovereign debt crisis. The difference between the actual and the counterfactual series is most striking in the case of the financial crisis where we observe that, without the negative contribution of MNCs' entry shocks, output growth would have entered positive territory already in 2009Q2, while in reality this happened one year later. This is in line with the empirical findings of Milesi-Ferretti and Tille (2014), who show that capital flows to emerging Europe remained at a very low level in the recovery phase of the crisis, which is in stark contrast with Latin America and emerging Asia, where capital inflows quickly resumed after the crisis.

6 Concluding Remarks

In this paper we have introduced a Schumpeterian creative destruction model of the productivity dynamics of a generic non-frontier small open economy in a DSGE model. The integration of Schumpeterian growth and DSGE is not straightforward, but it opens the door to a more rigorous and complete analysis of medium frequency effects of shocks in non-frontier countries than previously achievable only by DSGE or Schumpeterian growth models taken in isolation.

We have used this model to study the interplay between MNCs entry and domestic R&D in driving endogenous productivity in a non-frontier economy and used Hungarian data for our quantitative application. According to our estimated model, MNC's entry proved to be both boon and bane for Hungary since on the one hand it impacted positively economic growth, while on the other hand it was a significant source of business cycle amplification between 1995 and 2019. On what concerns the government subsidies, the R&D subsidy shock had almost no effect on the observed variables, which highlights the insignificant role played by this shock in driving domestic productivity or output growth in both the short and medium run.

One of the main contributions of this paper is the extension of the business cycle model with endogenous technology to the case of non-frontier SOEs that undertake R&D to catchup with the frontier but are at the same time exposed to MNCs' entry equipped with more advanced technology. Our estimation revealed that this is an empirically plausible model that can be used as a starting point to study the impact of MNCs entry on both the business cycle and the medium-to-long-term fluctuations in EMs.

Nevertheless, the proposed model purposefully abstracted from several important additional features that would increase the realism of the model and allow for more reliable estimation and policy evaluations, but that would also introduce complexity to a level that would render the exposition much less transparent. However, our setting could be extended in a relatively straightforward way in important directions. Most notably are two. One is the explicit introduction of a more complex government policy block with distortionary taxation, which would allow the simulation of government policies. Another is the introduction of nominal rigidities and trade, which would increase the realism of the model.

References

- Acemoglu, Daron, Ufuk Akcigit, Harun Alp, Nicholas Bloom, and William R. Kerr (2018). Innovation, Reallocation and Growth. American Economic Review, 108 (11): 3450-91. DOI: 10.1257/aer.20130470. URL: {https://www.aeaweb.org/articles?id=10. 1257/aer.20130470}.
- Ådám, Zoltán (2019). Explaining Orbán: A Political Transaction Cost Theory of Authoritarian Populism. Problems of Post-Communism, 66 (6): 385-401. DOI: 10.1080/10758216.
 2019.1643249. eprint: https://doi.org/10.1080/10758216.2019.1643249. URL: https://doi.org/10.1080/10758216.2019.1643249.
- Adams, Samuel and Eric Evans Osei Opoku (2015). Foreign direct investment, regulations and growth in sub-Saharan Africa. *Economic Analysis and Policy*, 47: 48-56. ISSN: 0313-5926. DOI: https://doi.org/10.1016/j.eap.2015.07.001. URL: https: //www.sciencedirect.com/science/article/pii/S0313592615000259.
- Adolfson, Malin, Stefan Laséen, Jesper Lindé, and Mattias Villani (2007). Bayesian estimation of an open economy DSGE model with incomplete pass-through. Journal of International Economics, 72(2): 481–511. DOI: https://doi.org/10.1016/j. jinteco.2007.01.003. URL: http://www.sciencedirect.com/science/article/ pii/S0022199607000219.
- (2008). Empirical Properties of Closed- and Open-Economy DSGE Models of the Euro Area. *Macroeconomic Dynamics*, 12 (S1): 2?19. DOI: 10.1017/S1365100507070101.
- Aghion, Philippe and Peter Howitt (1992). A Model of Growth through Creative Destruction. Econometrica, 60 (2): 323–351.
- Aguiar, M., S. Chatterjee, H. Cole, and Z. Stangebye (2016). Chapter 21 Quantitative Models of Sovereign Debt Crises. In: ed. by John B. Taylor and Harald Uhlig. Vol. 2. Handbook of Macroeconomics. Elsevier, p. 1697-1755. DOI: https://doi.org/10.1016/ bs.hesmac.2016.04.005. URL: https://www.sciencedirect.com/science/article/ pii/S1574004816300052.
- Aguiar, Mark and Gita Gopinath (2007). Emerging Market Business Cycles: The Cycle Is the Trend. Journal of Political Economy, 115(1): 69–102. DOI: 10.1086/511283. eprint: https://doi.org/10.1086/511283. URL: https://doi.org/10.1086/511283.
- Ahmed, Shaghil and Andrei Zlate (2014). Capital flows to emerging market economies: A brave new world? Journal of International Money and Finance, 48. Current account imbalances and international financial integration: 221-248. ISSN: 0261-5606. DOI: https://doi.org/10.1016/j.jimonfin.2014.05.015. URL: https://www.sciencedirect.com/science/article/pii/S0261560614000928.
- Aizenman, Joshua, Yothin Jinjarak, and Donghyun Park (2013). Capital Flows and Economic Growth in the Era of Financial Integration and Crisis, 1990–2010. Open Economies Review, 24 (3): 371–396.
- Albonico, Alice et al. (2019). The Global Multi-Country Model (GM): an Estimated DSGE Model for the Euro Area Countries. European Economy Discussion Papers 2019-102. Joint Research Centre, European Commission (Ispra site). DOI: 10.2765/254908.
- Alfaro, Laura, Areendam Chanda, Sebnem Kalemli-Ozcan, and Selin Sayek (2004). FDI and economic growth: the role of local financial markets. *Journal of International Economics*, 64(1): 89–112. ISSN: 0022-1996. DOI: https://doi.org/10.1016/S0022-

1996(03)00081-3. URL: https://www.sciencedirect.com/science/article/pii/ S0022199603000813.

- Alfaro, Laura, Areendam Chanda, Sebnem Kalemli-Ozcan, and Selin Sayek (2010). Does foreign direct investment promote growth? Exploring the role of financial markets on linkages. Journal of Development Economics, 91 (2): 242 -256. ISSN: 0304-3878. DOI: https: //doi.org/10.1016/j.jdeveco.2009.09.004. URL: http://www.sciencedirect. com/science/article/pii/S0304387809000947.
- Almfraji, Mohammad Amin and Mahmoud Khalid Almsafir (2014). Foreign Direct Investment and Economic Growth Literature Review from 1994 to 2012. Procedia - Social and Behavioral Sciences, 129. 2nd International Conference on Innovation, Management and Technology Research: 206 -213. ISSN: 1877-0428. DOI: https://doi.org/10.1016/j. sbspro.2014.03.668. URL: http://www.sciencedirect.com/science/article/pii/ S187704281402850X.
- An, Sungbae and Frank Schorfheide (2007). Bayesian Analysis of DSGE Models. *Econometric Reviews*, 26 (2-4): 113–172.
- Angeletos, George-Marios and Vasia Panousi (2011). Financial integration, entrepreneurial risk and global dynamics. *Journal of Economic Theory*, 146(3). Incompleteness and Uncertainty in Economics: 863-896. ISSN: 0022-0531. DOI: https://doi.org/10.1016/ j.jet.2011.02.001. URL: https://www.sciencedirect.com/science/article/pii/ S0022053111000172.
- Antalóczy, Katalin, Magdolna Sass, and Miklós Szanyi (2011). Policies for attracting foreign direct investment and enhancing its spillovers to indigenous firms: The case of Hungary. In: *Multinational Corporations and Local Firms in Emerging Economies*. Amsterdam University Press, p. 181–210. ISBN: 9789089642943. URL: http://www.jstor.org/stable/j.ctt46n0w0.10.
- Anzoategui, D., D.A. Comin, Gertler M., and J. Martinez (2019). Endogenous Technology Adoption and R&D as Sources of Business Cycle Persistence. American Economic Journal: Macroeconomics.
- Araujo, Juliana D., Antonio C. David, Carlos van Hombeeck, and Chris Papageorgiou (2017a). Joining the club? Procyclicality of private capital inflows in lower income developing economies. Journal of International Money and Finance, 70: 157–182. ISSN: 0261-5606. DOI: https://doi.org/10.1016/j.jimonfin.2016.08.006. URL: https: //www.sciencedirect.com/science/article/pii/S026156061630095X.
- Araujo, Juliana D., Antonio C. David, Carlos van Hombeeck, and Chris Papageorgiou (2017b). Non-FDI Capital Inflows in Low-Income Countries: Catching the Wave? *IMF Economic Review*, 65 (2): 426–465. DOI: 10.1057/s41308-016-0025-x. URL: https: //doi.org/10.1057/s41308-016-0025-x.
- Ates, Sina T. and Felipe Saffie (forthcoming). Fewer but Better : Sudden Stops, Firm Entry, and Financial Selection. *American Economic Journal: Macroeconomics*.
- Bačić, Katarina, Domagoj Račić, and Amina Ahec Šonje (2004). FDI and economic growth in Central and Eastern Europe: Is there a link? MPRA Paper 83136. University Library of Munich, Germany.
- Barrail, Zulma (2020). Business cycle implications of rising household credit market participation in emerging countries. *Journal of Economic Dynamics and Control*, 116: 103917.

ISSN: 0165-1889. DOI: https://doi.org/10.1016/j.jedc.2020.103917. URL: https: //www.sciencedirect.com/science/article/pii/S0165188920300853.

- Barseghyan, Levon, Francesca Molinari, Ted O'Donoghue, and Joshua C. Teitelbaum (2018). Estimating Risk Preferences in the Field. *Journal of Economic Literature*, 56 (2): 501–64. DOI: 10.1257/jel.20161148. URL: https://www.aeaweb.org/articles?id=10.1257/jel.20161148.
- Becker, Torbjörn et al. (2010). Whither growth in central and eastern Europe? Policy lessons for an integrated Europe. Vol. XI. Bruegel Blueprint Series.
- Benigno, Gianluca and Luca Fornaro (2012). Reserve Accumulation, Growth and Financial Crises. CEP Discussion Papers dp1161. Centre for Economic Performance, LSE.
- (2014). The Financial Resource Curse*. The Scandinavian Journal of Economics, 116(1): 58-86. DOI: https://doi.org/10.1111/sjoe.12047. eprint: https://onlinelibrary. wiley.com/doi/pdf/10.1111/sjoe.12047. URL: https://onlinelibrary.wiley. com/doi/abs/10.1111/sjoe.12047.
- (2017). Stagnation Traps. The Review of Economic Studies, 85(3): 1425–1470. ISSN: 1467-937X. DOI: 10.1093/restud/rdx063. URL: http://dx.doi.org/10.1093/restud/rdx063.
- Benigno, Gianluca, Nathan Converse, and Luca Fornaro (2015). Large capital inflows, sectoral allocation, and economic performance. *Journal of International Money and Finance*, 55 (C): 60–87. DOI: 10.1016/j.jimonfin.2015.0.
- Benigno, Gianluca, Luca Fornaro, and Martin Wolf (2020). *The Global Financial Resource Curse*. Working Papers 1165. Barcelona Graduate School of Economics.
- Benner, Thorsten and Wolfgang Reinicke (2017). Investing in Illiberalism. https://www.gppi.net/2017/07/28/investing-in-illiberalism.
- Bershidsky, Leonid (2017). How Western Capital Colonized Eastern Europe. https://www.bloomberg.com/opinion/articles/2017-09-12/how-western-capital-colonized-eastern-europe.
- Bianchi, Francesco, Howard Kung, and Gonzalo Morales (2018). Growth, slowdowns, and recoveries. Journal of Monetary Economics. ISSN: 0304-3932. DOI: https://doi.org/ 10.1016/j.jmoneco.2018.07.001. URL: http://www.sciencedirect.com/science/ article/pii/S0304393218303660.
- Bianchi, Javier (2011). Overborrowing and Systemic Externalities in the Business Cycle. American Economic Review, 101(7): 3400-3426. URL: https://ideas.repec.org/a/ aea/aecrev/v101y2011i7p3400-3426.html.
- Bijsterbosch, Martin and Marcin Kolasa (2010). FDI and productivity convergence in Central and Eastern Europe: an industry-level investigation. *Review of World Economics*, 145 (4): 689–712. DOI: 10.1007/s10290-009-0036-z.
- Blanchard, Olivier J., Mitali Das, Hamid Faruqee, Kristin J. Forbes, and Linda L. Tesar (2010). The Initial Impact of the Crisis on Emerging Market Countries [with Comments and Discussion]. Brookings Papers on Economic Activity: 263-323. ISSN: 00072303, 15334465. URL: http://www.jstor.org/stable/40930485.
- Boileau, Martin and Michel Normandin (2017). The price of imported capital and consumption fluctuations in emerging economies. *Journal of International Economics*, 108: 67–81. ISSN: 0022-1996. DOI: https://doi.org/10.1016/j.jinteco.2017.05.009. URL: https://www.sciencedirect.com/science/article/pii/S0022199617300648.

- Bolboaca, M. and S. Fischer (2019). Unraveling News: Reconciling Conflicting Evidences. Working Papers 19.02. Study Center Gerzensee.
- Bosworth, Barry P., Susan M. Collins, and Carmen M. Reinhart (1999). Capital Flows to Developing Economies: Implications for Saving and Investment. *Brookings Papers on Economic Activity*, 1999(1): 143–180. ISSN: 00072303, 15334465. URL: http://www. jstor.org/stable/2534664.
- Broner, Fernando, Tatiana Didier, Aitor Erce, and Sergio L. Schmukler (2013). Gross capital flows: Dynamics and crises. *Journal of Monetary Economics*, 60(1). Carnegie-NYU-Rochester Conference: 113–133. ISSN: 0304-3932. DOI: https://doi.org/10.1016/j. jmoneco.2012.12.004. URL: https://www.sciencedirect.com/science/article/ pii/S0304393212001675.
- Brooks, Stephen P. and Andrew Gelman (1998). General Methods for Monitoring Convergence of Iterative Simulations. *Journal of Computational and Graphical Statistics*, 7(4): 434-455. DOI: 10.1080/10618600.1998.10474787. eprint: https://www.tandfonline. com/doi/pdf/10.1080/10618600.1998.10474787. URL: https://www.tandfonline. com/doi/abs/10.1080/10618600.1998.10474787.
- Buera, Francisco J. and Yongseok Shin (2017). Productivity Growth and Capital Flows: The Dynamics of Reforms. American Economic Journal: Macroeconomics, 9(3): 147-85. DOI: 10.1257/mac.20160307. URL: https://www.aeaweb.org/articles?id=10.1257/mac. 20160307.
- Burstein, Ariel T. and Alexander Monge-Naranjo (2009). Foreign Know-How, Firm Control, and the Income of Developing Countries. *The Quarterly Journal of Economics*, 124 (1): 149–195.
- Caballero, Ricardo J., Emmanuel Farhi, and Pierre-Olivier Gourinchas (2008). An Equilibrium Model of "Global Imbalances" and Low Interest Rates. *American Economic Review*, 98 (1): 358-93. DOI: 10.1257/aer.98.1.358. URL: https://www.aeaweb.org/articles?id=10.1257/aer.98.1.358.
- Calvo, Guillermo A., Leonardo Leiderman, and Carmen M. Reinhart (1996). Inflows of Capital to Developing Countries in the 1990s. *Journal of Economic Perspectives*, 10(2): 123-139. DOI: 10.1257/jep.10.2.123. URL: https://www.aeaweb.org/articles?id= 10.1257/jep.10.2.123.
- Chang, Roberto and Andrés Fernández (2013). On the Sources of Aggregate Fluctuations in Emerging Economies. International Economic Review, 54(4): 1265-1293. DOI: https: //doi.org/10.1111/iere.12036. eprint: https://onlinelibrary.wiley.com/doi/ pdf/10.1111/iere.12036. URL: https://onlinelibrary.wiley.com/doi/abs/10. 1111/iere.12036.
- Chetty, Raj (2006). A New Method of Estimating Risk Aversion. *The American Economic Review*, 96 (5): 1821–1834. ISSN: 00028282. URL: http://www.jstor.org/stable/30034997.
- Christiano, L., M. Eichenbaum, and C.L. Evans (2005). Nominal Rigidities and the Dynamic Effects of a Shock to Monetary Policy. *Journal of Political Economy*, 113 (1): 1–45.
- Chu, Angus C., Guido Cozzi, and Silvia Galli (2014). Stage-dependent intellectual property rights. *Journal of Development Economics*, 106: 239-249. ISSN: 0304-3878. DOI: https://doi.org/10.1016/j.jdeveco.2013.10.005. URL: http://www.sciencedirect.com/science/article/pii/S030438781300148X.

- Comin, D. and M. Gertler (2006). Medium-Term Business Cycles. American Economic Review, 96 (3): 523–551.
- Comin, D. and B. Hobijn (2004). Cross-Country Technology Adoption: Making the Theories Face the Facts. *Journal of Monetary Economics*, 51 (1): 39–83.
- (2010). An Exploration of Technology Diffusion. *American Economic Review*, 100(5): 2031–2059.
- Comin, D., B. Hobijn, and E. Rovito (2008). Technology Usage Lags. *Journal of Economic Growth*, 13: 237–256.
- Comin, Diego and Marttí Mestieri (2018). If Technology Has Arrived Everywhere, Why Has Income Diverged? American Economic Journal: Macroeconomics, 10 (3): 137-78. DOI: 10.1257/mac.20150175. URL: http://www.aeaweb.org/articles?id=10.1257/mac. 20150175.
- Comin, Diego, Norman Loayza, Farooq Pasha, and Luis Serven (2014). Medium Term Business Cycles in Developing Countries. American Economic Journal: Macroeconomics, 6 (4): 209-45. DOI: 10.1257/mac.6.4.209. URL: https://www.aeaweb.org/articles?id=10.1257/mac.6.4.209.
- Cozzi, Guido and Giammario Impullitti (2016). Globalization and Wage Polarization. *Review* of *Economics and Statistics*, 98 (5): 984–1000. ISSN: 1530-9142. DOI: 10.1162/rest_a_ 00551. URL: http://dx.doi.org/10.1162/REST_a_00551.
- Cozzi, Guido, Beatrice Pataracchia, Philipp Pfeiffer, and Marco Ratto (2021). How Much Keynes and How Much Schumpeter? *European Economic Review*, 133: 103660. ISSN: 0014-2921. DOI: https://doi.org/10.1016/j.euroecorev.2021.103660. URL: https://www.sciencedirect.com/science/article/pii/S0014292121000131.
- De Loecker, Jan and Jan Eeckhout (2017). The Rise of Market Power and the Macroeconomic Implications. Working Paper 23687. National Bureau of Economic Research. DOI: 10. 3386/w23687. URL: http://www.nber.org/papers/w23687.
- (2018). *Global Market Power*. Working Paper 24768. National Bureau of Economic Research. DOI: 10.3386/w24768. URL: http://www.nber.org/papers/w24768.
- De Loecker, Jan and Frederic Warzynski (2012). Markups and Firm-Level Export Status. American Economic Review, 102(6): 2437-71. DOI: 10.1257/aer.102.6.2437. URL: http://www.aeaweb.org/articles?id=10.1257/aer.102.6.2437.
- De Vita, Glauco and Khine S. Kyaw (2008). Determinants of capital flows to developing countries: a structural VAR analysis. 35 (4): 304–322. DOI: 10.1108/01443580810895608. URL: https://doi.org/10.1108/01443580810895608.
- Dellis, Konstantinos, David Sondermann, and Isabel Vansteenkiste (2020). Drivers of genuine FDI inflows in advanced economies. *The Quarterly Review of Economics and Finance*. ISSN: 1062-9769. DOI: https://doi.org/10.1016/j.qref.2020.09.004. URL: https: //www.sciencedirect.com/science/article/pii/S1062976920301149.
- Dynare: Reference Manual Version 4 (2011). Dynare Working Papers 1. CEPREMAP.
- Eaton, Jonathan and Samuel Kortum (2002). Technology, Geography, and Trade. *Econometrica*, 70(5): 1741-1779. DOI: https://doi.org/10.1111/1468-0262.00352. eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/1468-0262.00352. URL: https://onlinelibrary.wiley.com/doi/abs/10.1111/1468-0262.00352.

- Eden, M. and H. Nguyen (2016). Reconciling Micro- and Macro-Based Estimates of Technology Adoption Lags in a Model of Endogenous Technology Adoption. In: *Understanding* the Income and Efficiency Gap in Latin America and the Caribbean. Chap. 3, p. 57–72.
- Eder, Marton (2021). Orban Eyes Telecoms as Next Target for Hungarian Ownership Drive. https://www.bloombergquint.com/business/orban-eyes-telecoms-as-next-target-for-hungarian-ownership-drive.
- Eggertsson, Gauti B., Neil R. Mehrotra, Sanjay R. Singh, and Lawrence H. Summers (2016). A Contagious Malady? Open Economy Dimensions of Secular Stagnation. *IMF Economic Review*, 64 (4): 581–634. ISSN: 20414161, 2041417X. URL: http://www.jstor.org/stable/45212123.
- Eurostat (2018). Foreign direct investment intensity ratios. http://ec.europa.eu/ eurostat/statistics-explained/index.php/Foreign_direct_investment_-_intensity_ratios, accessed 06.06.2018.
- Fair, Ray C. (2020). Variable mismeasurement in a class of DSGE models: Comment. Journal of Macroeconomics, 66: 103242. ISSN: 0164-0704. DOI: https://doi.org/10.1016/j. jmacro.2020.103242. URL: http://www.sciencedirect.com/science/article/pii/ S0164070420301683.
- Feenstra, Robert C., Robert Inklaar, and Marcel P. Timmer (2015). The Next Generation of the Penn World Table. American Economic Review, 105 (10): 3150-82. DOI: 10.1257/ aer.20130954. URL: http://www.aeaweb.org/articles?id=10.1257/aer.20130954.
- Fernald, J. (2014). A Quarterly, Utilization-Adjusted Series on Total Factor Productivity. *Federal Reserve Bank of San Franciso*, (2012-19). Working Paper Series.
- Fidrmuc, Jarko and Reiner Martin (2011). FDI, trade and growth in CESEE countries. Focus on European Economic Integration, (1): 70–89.
- Fiorito, R. and G. Zanella (2012). The Anatomy of the Aggregate Labor Supply Elasticity. *Review of Economic Dynamics*, 15 (2): 171–187.
- Forbes, Kristin J. and Francis E. Warnock (2012). Capital flow waves: Surges, stops, flight, and retrenchment. *Journal of International Economics*, 88 (2). NBER Global: 235–251. ISSN: 0022-1996. DOI: https://doi.org/10.1016/j.jinteco.2012.03.006. URL: https://www.sciencedirect.com/science/article/pii/S0022199612000566.
- Galí, Jordi, Frank Smets, and Rafael Wouters (2012). Unemployment in an Estimated New Keynesian Model. *NBER Macroeconomics Annual*, 26 (1): 329–360. DOI: 10.1086/663994.
- Gandelman, Nestor and Ruben Hernandez-Murillo (2015). Risk Aversion at the Country Level. *Review, Federal Reserve Bank of St. Louis*, 97 (1): 53–66.
- Gelain, Paolo and Dmitry Kulikov (2011). An Estimated Dynamic Stochastic General Equilibrium Model with Financial Frictions for Estonia. *Eastern European Economics*, 49 (5): 97–120. DOI: 10.2753/EEE0012-8775490505. eprint: https://doi.org/10.2753/EEE0012-8775490505. URL: https://doi.org/10.2753/EEE0012-8775490505.
- Georgiev, Yordan, Piroska Nagy-Mohacsi, and Alexander Plekhanov (2017). Structural Reform and Productivity Growth in Emerging Europe and Central Asia. Research Paper Series 2. LSE Institute of Global Affairs.
- Ghosh, Atish R., Mahvash S. Qureshi, Jun Il Kim, and Juan Zalduendo (2014). Surges. Journal of International Economics, 92 (2): 266–285. ISSN: 0022-1996. DOI: https://

doi.org/10.1016/j.jinteco.2013.12.007. URL: https://www.sciencedirect.com/ science/article/pii/S002219961300144X.

- Gomme, P., B. Ravikumar, and P. Rupert (2015). Secular Stagnation and Returns on Capital. *Economic Synopses*, (19).
- Gopinath, Gita, Sebnem Kalemli-Ozcan, Loukas Karabarbounis, and Carolina Villegas-Sanchez (2017). Capital Allocation and Productivity in South Europe. *The Quarterly Journal* of Economics, 132(4): 1915–1967. ISSN: 0033-5533. DOI: 10.1093/qje/qjx024. URL: https://doi.org/10.1093/qje/qjx024.
- Gourinchas, Pierre-Olivier and Olivier Jeanne (2013). Capital Flows to Developing Countries: The Allocation Puzzle. *The Review of Economic Studies*, 80 (4): 1484–1515. ISSN: 1467-937X. DOI: 10.1093/restud/rdt004. URL: http://dx.doi.org/10.1093/restud/ rdt004.
- Greenwood, J., Z. Hercowitz, and G. Huffman (1988). Investment, Capacity Utilization, and the Real Business Cycle. *American Economic Review*, 78 (3): 402–417.
- Guerrón-Quintana, Pablo and Ryo Jinnai (2014). Liquidity, Trends, and the Great Recession. Working Paper 14-24. FRB of Philadelphia. DOI: 10.2139/ssrn.2494398. URL: http: //dx.doi.org/10.2139/ssrn.2494398.
- Hall, R.E. (1988). The Relation Between Price and Marginal Cost in U.S. Industry. *Journal* of Political Economy, 96 (5): 921–947.
- Havranek, Tomas, Marek Rusnak, and Anna Sokolova (2017). Habit formation in consumption: A meta-analysis. *European Economic Review*, 95: 142–167. ISSN: 0014-2921. DOI: https://doi.org/10.1016/j.euroecorev.2017.03.009. URL: http://www.sciencedirect.com/science/article/pii/S0014292117300569.
- Helpman, Elhanan, Marc J. Melitz, and Stephen R. Yeaple (2004). Export Versus FDI with Heterogeneous Firms. American Economic Review, 94(1): 300-316. DOI: 10.1257/ 000282804322970814. URL: https://www.aeaweb.org/articles?id=10.1257/ 000282804322970814.
- Hlavacek, Petr and Beata Bal-Domanska (2016). Impact of foreign direct investment on economic growth in Central and Eastern European countries. *Engineering Economics*, 27 (3): 294–303.
- Houssa, Romain, Jolan Mohimont, and Chris Otrok (2019). A model for international spillovers to emerging markets. Working Paper Research 370. National Bank of Belgium. URL: https://ideas.repec.org/p/nbb/reswpp/201904-370.html.
- Howitt, Peter and Philippe Aghion (1998). Capital Accumulation and Innovation as Complementary Factors in Long-Run Growth. *Journal of Economic Growth*, 3(2): 111–130. ISSN: 1573-7020. DOI: 10.1023/A:1009769717601. URL: https://doi.org/10.1023/A: 1009769717601.
- Hunya, G. (2017). Condemned to be Left Behind? Can Central and Eastern Europe Emerge from Its Low-wage Model? In: ed. by B. Galgóczi and J. Drahokoupil. ETUI aisbl, Brussels. Chap. Conditions for an investment revival in Central and Eastern Europe, p. 25– 46.
- Iamsiraroj, Sasi and Mehmet Ali Ulubaşoğlu (2015). Foreign direct investment and economic growth: A real relationship or wishful thinking? *Economic Modelling*, 51: 200 -213. ISSN: 0264-9993. DOI: https://doi.org/10.1016/j.econmod.2015.08.009. URL: http: //www.sciencedirect.com/science/article/pii/S0264999315002138.

- Irarrazabal, Alfonso, Andreas Moxnes, and Luca David Opromolla (2013). The Margins of Multinational Production and the Role of Intrafirm Trade. *Journal of Political Economy*, 121 (1): 74–126. DOI: 10.1086/669877. eprint: https://doi.org/10.1086/669877. URL: https://doi.org/10.1086/669877.
- Iskrev, Nikolay (2010). Local identification in DSGE models. Journal of Monetary Economics, 57 (2): 189 -202. ISSN: 0304-3932. DOI: https://doi.org/10.1016/j.jmoneco.2009.12.007. URL: http://www.sciencedirect.com/science/article/pii/S0304393209001883.
- Ivashchenko, Sergey and Willi Mutschler (2020). The effect of observables, functional specifications, model features and shocks on identification in linearized DSGE models. *Economic Modelling*, 88: 280 -292. ISSN: 0264-9993. DOI: https://doi.org/10.1016/j. econmod.2019.09.039. URL: http://www.sciencedirect.com/science/article/ pii/S0264999319307850.
- Jaccard, Ivan and Frank Smets (2020). Structural asymmetries and financial imbalances in the Eurozone. Review of Economic Dynamics, 36: 73-102. ISSN: 1094-2025. DOI: https: //doi.org/10.1016/j.red.2019.08.002. URL: https://www.sciencedirect.com/ science/article/pii/S109420251830156X.
- Jaimovich, N. and S. Rebelo (2009). Can News About the Future Drive the Business Cycle? American Economic Review, 99 (4): 1097–1118.
- Jakab, Zolán and István Kónya (2016). An Open Economy DSGE Model with Search-and-Matching Frictions: The Case of Hungary. *Emerging Markets Finance and Trade*, 52 (7): 1606–1626. DOI: 10.1080/1540496X.2014.1000174. eprint: https://doi.org/10. 1080/1540496X.2014.1000174. URL: https://doi.org/10.1080/1540496X.2014. 1000174.
- Javorcik, Beata S. (2015). Does FDI Bring Good Jobs to Host Countries? World Bank Research Observer, 30(1): 74–94.
- Justiniano, Alejandro and Bruce Preston (2010). Monetary policy and uncertainty in an empirical small open-economy model. Journal of Applied Econometrics, 25(1): 93-128. DOI: https://doi.org/10.1002/jae.1153. eprint: https://onlinelibrary.wiley. com/doi/pdf/10.1002/jae.1153. URL: https://onlinelibrary.wiley.com/doi/ abs/10.1002/jae.1153.
- Justiniano, Alejandro, Giorgio E. Primiceri, and Andrea Tambalotti (2010). Investment shocks and business cycles. *Journal of Monetary Economics*, 57 (2): 132–145.
- Justiniano, Alejandro, Giorgio Primiceri, and Andrea Tambalotti (2011). Investment Shocks and the Relative Price of Investment. *Review of Economic Dynamics*, 14(1): 101–121. DOI: 10.1016/j.red.2010.08.004.
- Kalotay, Kálmán (2017). Post-crisis Crossroads for FDI in CEE. In: Foreign Direct Investment in Central and Eastern Europe: Post-crisis Perspectives. Ed. by Balázs Szent-Iványi. Cham: Springer International Publishing, p. 23–49. ISBN: 978-3-319-40496-7. DOI: 10.1007/978-3-319-40496-7_2. URL: https://doi.org/10.1007/978-3-319-40496-7_2.
- Kaminsky, Graciela L., Carmen M. Reinhart, and Carlos A. Végh (2004). When It Rains, It Pours: Procyclical Capital Flows and Macroeconomic Policies. *NBER Macroeconomics Annual*, 19: 11–53. DOI: 10.1086/ma.19.3585327. eprint: https://doi.org/10.1086/ ma.19.3585327. URL: https://doi.org/10.1086/ma.19.3585327.

- Karabarbounis, L. and B. Neiman (2013). The Global Decline of the Labor Share. *The Quarterly Journal of Economics*, 129 (1): 61–103. ISSN: 1531-4650. DOI: 10.1093/qje/qjt032. URL: http://dx.doi.org/10.1093/qje/qjt032.
- Keane, M. and R. Rogerson (2012). Micro and Macro Labor Supply Elasticities: A Reassessment of Conventional Wisdom. *Journal of Economic Literature*, 50 (2): 464–76.
- Khan, H. and J. Tsoukalas (2012). The Quantitative Importance of News Shocks in Estimated DSGE Models. *Journal of Money, Credit and Banking*, 44(8): 1535–1561.
- Kim, Yoonbai (2000). Causes of capital flows in developing countries. Journal of International Money and Finance, 19 (2): 235-253. ISSN: 0261-5606. DOI: https://doi.org/10.1016/ S0261-5606(00)00001-2. URL: https://www.sciencedirect.com/science/article/ pii/S0261560600000012.
- King, R., C. Plosser, and S. Rebelo (1988). Production, Growth and Business Cycles: I. The Basic Neoclassical Model. *Journal of Monetary Economics*, 21 (2-3): 195–232.
- Klaudt, Kent (1995). Hungary after the Revolution: Privatization, Economic Ideology and the False Promise of the Free Market. *Law and Inequality*, 13: 303.
- Klette, Tor Jakob and Samuel Kortum (2004). Innovating Firms and Aggregate Innovation. Journal of Political Economy, 112 (5): 986–1018. DOI: 10.1086/422563.
- Komunjer, Ivana and Serena Ng (2011). Dynamic identification of dynamic stochastic general equilibrium models. *Econometrica*, 79 (6): 1995–2032.
- Kónya, István and Franklin Maduko (2020). Interest premium and external position: A state dependent approach. Journal of International Financial Markets, Institutions and Money, 66: 101192. ISSN: 1042-4431. DOI: https://doi.org/10.1016/j.intfin. 2020.101192. URL: http://www.sciencedirect.com/science/article/pii/S1042443120300767.
- Kose, Ayhan M., Eswar S. Prasad, and Marco E. Terrones (2009). Does openness to international financial flows raise productivity growth? *Journal of International Money and Finance*, 28 (4): 554–580.
- Kung, Howard and Lukas Schmid (2015). Innovation, Growth, and Asset Prices. *The Journal* of Finance, 70 (3): 1001–1037. ISSN: 0022-1082. DOI: 10.1111/jofi.12241. URL: http://dx.doi.org/10.1111/jofi.12241.
- Lane, Philip R. (2015). International Financial Flows in Low-Income Countries. Pacific Economic Review, 20(1): 49-72. DOI: https://doi.org/10.1111/1468-0106.12094. eprint: https://onlinelibrary.wiley.com/doi/pdf/10.1111/1468-0106.12094. URL: https://onlinelibrary.wiley.com/doi/abs/10.1111/1468-0106.12094.
- Levenko, Natalia, Kaspar Oja, and Karsten Staehr (2017). Total Factor Productivity Growth in Central and Eastern Europe before, During and After the Global Financial Crisis. Working Paper Series 8. Eesti Pank.
- Lucas, Robert E. (1978). On the Size Distribution of Business Firms. The Bell Journal of Economics, 9(2): 508-523. ISSN: 0361915X. URL: http://www.jstor.org/stable/ 3003596.
- McGrattan, Ellen (2012). Transition to FDI Openness: Reconciling Theory and Evidence. Review of Economic Dynamics, 15(4): 437–458. DOI: 10.1016/j.red.2012.07.004.
- McGrattan, Ellen R. and Edward C. Prescott (2009). Openness, technology capital, and development. *Journal of Economic Theory*, 144 (6): 2454–2476.

- Melitz, Marc J. (2003). The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity. *Econometrica*, 71 (6): 1695–1725. DOI: https://doi.org/10. 1111/1468-0262.00467. eprint: https://onlinelibrary.wiley.com/doi/pdf/10. 1111/1468-0262.00467. URL: https://onlinelibrary.wiley.com/doi/abs/10. 1111/1468-0262.00467.
- Mello LR, Jr de (1999). Foreign direct investment-led growth: evidence from time series and panel data. Oxford Economic Papers, 51(1): 133-151. ISSN: 0030-7653. DOI: 10.1093/ oep/51.1.133. eprint: https://academic.oup.com/oep/article-pdf/51/1/133/ 9901738/133.pdf. URL: https://doi.org/10.1093/oep/51.1.133.
- Mendoza, Enrique G, Vincenzo Quadrini, and Jose-Victor Rios-Rull (2009). Financial Integration, Financial Development, and Global Imbalances. *Journal of Political Economy*, 117 (3): 371-416. ISSN: 00223808, 1537534X. URL: http://www.jstor.org/stable/10. 1086/599706.
- Milesi-Ferretti, Gian-Maria and Cédric Tille (2014). The great retrenchment: international capital flows during the global financial crisis. *Economic Policy*, 26 (66): 289–346. ISSN: 0266-4658. DOI: 10.1111/j.1468-0327.2011.00263.x. URL: https://doi.org/10.1111/j.1468-0327.2011.00263.x.
- Moran, Patrick and Albert Queralto (2018). Innovation, productivity, and monetary policy. Journal of Monetary Economics, 93. Carnegie-Rochester-NYU Conference on Public Policy held at the Stern School of Business at New York University: 24-41. ISSN: 0304-3932. DOI: https://doi.org/10.1016/j.jmoneco.2017.10.006. URL: https: //www.sciencedirect.com/science/article/pii/S0304393217301216.
- Neumeyer, Pablo A. and Fabrizio Perri (2005). Business cycles in emerging economies: the role of interest rates. *Journal of Monetary Economics*, 52 (2): 345-380. ISSN: 0304-3932. DOI: https://doi.org/10.1016/j.jmoneco.2004.04.011. URL: https://www.sciencedirect.com/science/article/pii/S0304393205000036.
- Nuño, Galo (2011). Optimal research and development and the cost of business cycles. *Journal of Economic Growth*, 16 (3): 257. ISSN: 1573-7020. DOI: 10.1007/s10887-011-9063-4. URL: https://doi.org/10.1007/s10887-011-9063-4.
- OECD (2002). Measuring Globalisation: The Role of Multinationals in OECD Economies 2001, Volume I, Manufacturing Sector, p. 414. DOI: https://doi.org/https://doi. org/10.1787/9789264095939-en-fr. URL: https://www.oecd-ilibrary.org/ content/publication/9789264095939-en-fr.
- Peterman, William B. (2016). Reconciling Micro and Macro Estimates of the Frisch Labor Supply Elasticity. *Economic Inquiry*, 54 (1): 100–120.
- Prades, Elvira and Katrin Rabitsch (2012). Capital liberalization and the US external imbalance. Journal of International Economics, 87 (1). Symposium on the Global Dimensions of the Financial Crisis: 36-49. ISSN: 0022-1996. DOI: https://doi.org/10.1016/j. jinteco.2011.12.002. URL: https://www.sciencedirect.com/science/article/ pii/S0022199611001656.
- Qu, Zhongjun and Denis Tkachenko (2012). Identification and frequency domain quasimaximum likelihood estimation of linearized dynamic stochastic general equilibrium models. *Quantitative Economics*, 3(1): 95–132.
- Queralto, Albert (2020). A model of slow recoveries from financial crises. *Journal of Monetary Economics*, 114: 1–25. ISSN: 0304-3932. DOI: https://doi.org/10.1016/j.jmoneco.

2019.03.008. URL: https://www.sciencedirect.com/science/article/pii/ S0304393219300546.

- Radosevic, Slavo and Esin Yoruk (2015). A New Metrics of Technology Upgrading: The Central And East European Countries in a Comparative Perspective. GRINCOH Working Paper Series 3.04. University College London.
- (2016). Why do we need a theory and metrics of technology upgrading? Asian Journal of Technology Innovation, 24 (sup1): 8–32. ISSN: 2158-6721. DOI: 10.1080/19761597.
 2016.1207415. URL: http://dx.doi.org/10.1080/19761597.2016.1207415.
- (2018). Technology upgrading of middle income economies: A new approach and results. *Technological Forecasting and Social Change*, 129: 56-75. ISSN: 0040-1625. DOI: https: //doi.org/10.1016/j.techfore.2017.12.002. URL: http://www.sciencedirect. com/science/article/pii/S0040162516308368.
- Ramondo, Natalia (2014). A quantitative approach to multinational production. Journal of International Economics, 93(1): 108-122. ISSN: 0022-1996. DOI: https://doi.org/10. 1016/j.jinteco.2014.01.004. URL: https://www.sciencedirect.com/science/ article/pii/S0022199614000063.
- Ramondo, Natalia and Andrés Rodríguez-Clare (2013). Trade, Multinational Production, and the Gains from Openness. *Journal of Political Economy*, 121(2): 273–322. DOI: 10.1086/670136. eprint: https://doi.org/10.1086/670136. URL: https://doi.org/10.1086/670136.
- Real Business Cycles in Emerging Countries? (2010). American Economic Review, 100(5): 2510-31. DOI: 10.1257/aer.100.5.2510. URL: https://www.aeaweb.org/articles? id=10.1257/aer.100.5.2510.
- Reinhart, Carmen and Vincent Reinhart (2009). Capital Flow Bonanzas: An Encompassing View of the Past and Present. In: *NBER International Seminar on Macroeconomics 2008*. University of Chicago Press, p. 9–62. URL: http://www.nber.org/chapters/c8229.
- Reis, Ricardo (2013). The Portuguese Slump and Crash and the Euro Crisis. *Brookings* Papers on Economic Activity: 143-193. ISSN: 00072303, 15334465. URL: http://www. jstor.org/stable/23594865.
- Ríos-Rull, José-Víctor, Frank Schorfheide, Cristina Fuentes-Albero, Maxym Kryshko, and Raül Santaeulàlia-Llopis (2012). Methods versus substance: Measuring the effects of technology shocks. *Journal of Monetary Economics*, 59 (8): 826-846. DOI: https://doi. org/10.1016/j.jmoneco.2012.10.008. URL: http://www.sciencedirect.com/ science/article/pii/S0304393212001134.
- Roeger, Werner, Janos Varga, and Jan in 't Veld (2008). Structural Reforms in the EU: A simulation-based analysis using the QUEST model with endogenous growth. European Economy - Economic Papers 351. Directorate General Economic and Financial Affairs (DG ECFIN), European Commission.
- (2016). *Financial crisis and TFP growth in the EA*. Discussion Paper. Directorate General Economic and Financial Affairs (DG ECFIN), European Commission.
- Romer, Paul M. (1990). Endogenous Technological Change. *Journal of Political Economy*, 98 (5, Part 2): S71–S102. ISSN: 1537-534X. DOI: 10.1086/261725. URL: http://dx.doi.org/10.1086/261725.

- Saffie, Felipe, Liliana Varela, and Kei-Mu Yi (2020). The Micro and Macro Dynamics of Capital Flows. NBER Working Papers 27371. National Bureau of Economic Research, Inc.
- Sandri, Damiano (2014). Growth and Capital Flows with Risky Entrepreneurship. American Economic Journal: Macroeconomics, 6(3): 102-23. DOI: 10.1257/mac.6.3.102. URL: https://www.aeaweb.org/articles?id=10.1257/mac.6.3.102.
- Sarnyai, Gábor (2018). Hungary Has Given 86.5 Billion to Multinational Companies This Year to Create Jobs. https://hungarytoday.hu/hungary-has-given-86-5-billion-to-multinationalcompanies-this-year-to-create-jobs/.
- Sass, Magdolna (2004). FDI in Hungary the first mover's advantage and disadvantage. EIB Papers 8/2004. European Investment Bank, Economics Department.
- Schmitt-Grohé, S. and M. Uribe (2012). What's News in Business Cycles. *Econometrica*, 80 (6): 2733?2764.
- Schmitt-Grohé, Stephanie and Martín Uribe (2003). Closing small open economy models. Journal of International Economics, 61(1): 163 -185. ISSN: 0022-1996. DOI: https: //doi.org/10.1016/S0022-1996(02)00056-9. URL: http://www.sciencedirect. com/science/article/pii/S0022199602000569.
- Siena, Daniele (2021). The Euro Area Periphery and Imbalances: Is it an Anticipation Story? *Review of Economic Dynamics*, 40: 278–308. DOI: 10.1016/j.red.2020.09.006.
- Smets, Frank and Raf Wouters (2003). An Estimated Dynamic Stochastic General Equilibrium Model of the Euro Area.
- Smets, Frank and Rafael Wouters (2007). Shocks and Frictions in US Business Cycles: A Bayesian DSGE Approach. American Economic Review, 97 (3): 586–606. DOI: 10.1257/ aer.97.3.586. URL: http://www.aeaweb.org/articles?id=10.1257/aer.97.3.586.
- U.S. Department of State (2019). 2019 Investment Climate Statements: Hungary. https://www.state.gov/reports/2019-investment-climate-statements/hungary/.
- Valentinyi, Ákos (2012). EEAG Report on the European Economy. In: ed. by Lars Calmfors, Giancarlo Corsetti, John Hassler, Gilles Saint-Paul, Hans-Werner Sinn, Jan-Egbert Sturm, Ákos Valentinyi, and Xavier Vives. CESifo Group Munich, Munich. Chap. Chapter 5: The Hungarian Crisis, p. 115–130.

A MNCs in the CSEE Region

As it can be observed in Table 4, foreign enterprises represent approximately three percent of the total number of firms in all groups. However, the share of foreign control in terms of either turnover, production value, or value added, is different between the groups. Regardless of the measure employed, foreign control is higher in the CSEE than in the EU15. The share of turnover under foreign control is about 38 percent in the CSEE, but slightly more than 30 percent in the EU15. The discrepancy is even higher when comparing the foreign penetration in terms of value added, but highest in terms of production value, the share for the CSEE being 8.6 percentage points higher.

When comparing the performance of foreign versus domestic enterprises in the three groups of countries based on productivity, we observe that labor productivity is larger in foreign firms than in their domestic counterparts regardless of the measure employed or the region considered.⁶⁶ The greatest disparity in labor productivity between foreign and domestic enterprises, using gross value added per person employed as a proxy, is registered in the CSEE region. However, when considering the wage adjusted labor productivity, measured as gross value added divided by personnel costs, the difference between the productivity in foreign-controlled and domestic firms is largest in the EU15. The reason for this could be the higher wages paid by foreign-controlled companies relative to those paid by the local firms, the gap being larger in the CSEE than in the EU15.

	CSEE	EU15	EU28
Foreign enterprises	3.1	3.0	2.9
Foreign penetration based on turnover	38.0	30.2	32.3
Foreign penetration based on production value	38.5	29.9	32.5
Foreign penetration based on value added	34.3	26.0	28.6
Relative apparent labor productivity	166.6	161.4	164.3
Relative wage adjusted labor productivity	117.7	121.9	120.0

Table 4: MNCs at the level of 2014. Comparison of country groups

The ratios for individual countries reported in Table 5 show that there is a wide variation within the CSEE region. Most foreign-controlled enterprises are registered in Latvia, 7.1 percent of the total number of companies, and Romania, 6.5 percent, while the least are in Poland, 0.5 percent of total, and Slovak Republic, 0.8 percent. Interestingly, foreign penetration in terms of either turnover, production value, or value added, in the countries with smaller shares of foreign-controlled enterprises is relatively high, which implies that fewer multinational companies generate a large turnover, production value, or value added. For example, in Poland these three foreign penetration indicators are close to 30 percent and are similar to the respective indicators for Slovenia where the number of foreign companies

⁶⁶The labor productivity of domestic firms used in this analysis is in fact the labor productivity of all firms registered in the countries included in each group. However, as foreign firms represent a small share of the total number of firms, they may influence very little the average value of labor productivity. In any case, this would imply a small downward bias of the ratios reported.

represents 5.1 percent of the total. The highest penetration of foreign enterprises is observed in Hungary for which any of the indicators is above 50 percent even though the number of foreign enterprises in the economy is close to the CSEE average.

What is more important however is to compare the foreign penetration indicators used. Hunya (2017) shows that a direct comparison of these foreign penetration indicators provides important information concerning the activity of foreign affiliates in particular countries. For example, the author points to the fact that the Slovak Republic and Hungary have a similar share of foreign penetration in terms of production value, which is above 55 percent. Nevertheless, the indicator in terms of value added is more than 9 percentage points higher in Hungary than in the Slovak Republic. Hunya (2017) states that this in an indicator of the fact that the Slovak Republic is specialized in assembly work with a high imported content. In contrast, in the case of Poland the share of foreign penetration in terms of value added is higher than in production which makes Poland a more diverse economy, with much domestic sourcing.

Table 5: MNCs at the level of 2014.	Comparison of in	idividual countries
-------------------------------------	------------------	---------------------

	Bulgaria	Croatia	Czech Rep.	Estonia	Hungary	Latvia
Foreign enterprises	3.5	2.8	1.3	1.2	3.6	7.1
Foreign penetration based on turnover	33.9	29.0	46.7	26.9	52.8	39.3
Foreign penetration based on production value	38.7	28.1	48.4	29.9	57.4	29.1
Foreign penetration based on value added	33.0	24.9	42.3	26.4	52.7	31.1
Employees in foreign enterprises	18.2	:	33.4	23.7	29.9	19.2

	Lithuania	Poland	Romania	Slovakia	Slovenia	CSEE
Foreign enterprises - % of total	2.1	0.5	6.5	0.8	5.1	3.1
Foreign penetration based on turnover	32.5	29.6	48.1	49.9	29.1	38.0
Foreign penetration based on production value	34.3	27.9	47.3	55.3	26.6	38.5
Foreign penetration based on value added	26.5	29.1	43.9	43.4	24.1	34.3
Employees in foreign enterprises - $\%$ of total	6.3	:	27.5	:	:	:

Note that ":" stands for missing information.

Table 6: MNCs in Hungary as a Percentage of the Business Sector

	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017	2018	Average
Foreign enterprises	3.3	3.3	3.3	3.4	3.4	3.5	3.6	3.2	2.9	2.7	2.6	3.2
Foreign penetration based on turnover	50.1	51.7	51.6	53.1	53.3	53.1	52.8	53.2	51.4	49.9	48.8	51.7
Foreign penetration based on production value	52.8	53.6	54.9	57.1	57.5	57.1	57.4	58.4	57.8	56.3	54.6	56.1
Foreign penetration based on value added	47	49.1	49.3	51.9	51.7	52.2	52.7	52.5	51.4	50	47.6	50.5
Employees in foreign enterprises	27.1	27.7	28.4	29.8	29.7	29.9	29.9	29.5	29.3	29.8	30.1	29.2

B Competitive Equilibrium

The competitive equilibrium for this model is defined by a set of prices and allocations such that, taking the initial conditions and the stochastic processes for the exogenous variables as given, household's allocation satisfies the optimal choices defined by the first order conditions; firms' allocations maximize profits; all markets clear; and budget constraints hold with equality.

The small open economy we are analyzing is always growing at the trend growth rate of exogenous frontier technology. Therefore in order to solve the model in a way manageable to analyze medium frequencies, we need to stationarize the trending variables. We will denote both stationary and stationarized variables with lowercase letters. For example, domestic productivity A_t , which eventually grows at the trend rate of A_t^{\max} , is stationarized by dividing it by A_t^{\max} , obtaining $a_t = \frac{A_t}{A_t^{\max}}$. We proceed similarly for all variables asymptotically growing like A_t , most notably output, consumption, wage rate, investments, physical capital, foreign bonds, trade balance, etc. Finally, variables with no trend, such as the interest rate, the rental rate of capital, and the probability of innovation by incumbents or MNCs, etc. have not been further stationarized. In equations in which variables appear at different dates, it will be necessary to deflate stationarized past variables and inflate stationarized forward-looking variables by the trend growth rate. We further deflate the nominal variables using P_t as deflator as done in the equations derived from solving the household's problem, but given the normalization $P_t = 1$ this brings no further changes. With all this is mind, the reader can now easily interpret the meaning of the stationarized model equations listed below.

The growth rate of frontier technology is:

$$g_t = g_{\text{exo}} + \epsilon_t^g.$$

The law of motion of s_t can be written as:

$$s_{t} = \left[c_{t} \exp\left(-g_{t}\right)^{\frac{1-\chi}{\chi}} - hc_{t-1} \exp\left(-g_{t}\right)^{\frac{1}{\chi}}\right]^{\chi} s_{t-1}^{1-\chi}.$$

The Euler Equation is:

$$\lambda_t = \mathbb{E}_t \beta \lambda_{t+1} \exp\left(-g_{t+1}\right)^{\sigma} (1+r_t).$$

The first order condition with respect to consumption is given by:

$$\lambda_{t} = \exp(\epsilon_{t}^{\text{beta}}) \left[c_{t} - hc_{t-1} \exp\left(-g_{t}\right) - \omega \theta^{-1} l_{t}^{\theta} s_{t} \right]^{-\sigma} - \beta h \mathbb{E}_{t} \exp(\epsilon_{t+1}^{\text{beta}}) \left[c_{t+1} \exp\left(g_{t+1}\right) - hc_{t} - \omega \theta^{-1} l_{t+1}^{\theta} s_{t+1} \exp\left(g_{t+1}\right) \right]^{-\sigma} - \pi_{t} \chi \left(c_{t} \exp\left(g_{t}\right) - hc_{t-1} \right)^{\chi - 1} s_{t-1}^{1-\chi} + \beta h \chi \mathbb{E}_{t} \pi_{t+1} \exp\left(-g_{t+1}\right)^{\sigma} \left(c_{t+1} \exp\left(g_{t+1}\right) - hc_{t} \right)^{\chi - 1} s_{t}^{1-\chi}.$$

The labor supply equation in stationarized form reads:

$$\exp(\epsilon_t^{\text{beta}}) \left[c_t - hc_{t-1} \exp\left(-g_t\right) - \omega \theta^{-1} l_t^{\theta} s_t \right]^{-\sigma} \omega(l_t)^{\theta-1} s_t = w_t \lambda_t.$$

The first order condition with respect to s_t is given by:

$$\pi_{t} = \exp(\epsilon_{t}^{\text{beta}}) \left[c_{t} - hc_{t-1} \exp(-g_{t}) - \omega \theta^{-1} l_{t}^{\theta} s_{t} \right]^{-\sigma} \omega \theta^{-1} l_{lt}^{\theta} + \beta (1-\chi) \mathbb{E}_{t} \pi_{t+1} \exp(-g_{t+1})^{\sigma} \left(c_{t+1} \exp(g_{t+1}) - hc_{t} \right)^{\chi} s_{t}^{-\chi}.$$

The capital accumulation equation is:

$$k_t = \exp(\epsilon_t^{\text{MEI}}) i_t \left[1 - \eta_I \left(\frac{i_t}{i_{t-1}} \right) \right] + (1 - \delta) k_{t-1} \exp\left(-g_t \right).$$

The price of capital equation becomes:

$$p_t^K \lambda_t = \mathbb{E}_t \beta \lambda_{t+1} \exp\left(-g_{t+1}\right) \left[r_{t+1}^K + p_{t+1}^K (1-\delta)\right],$$

while optimal investment choice now implies:

$$1 = \exp(\epsilon_t^{\text{MEI}}) p_t^K \left[1 - \frac{\gamma_K}{2} \left(\frac{i_t}{i_{t-1}} \exp\left(g_t\right) - exp\left(g_I\right) \right)^2 - \gamma_K \frac{i_t}{i_{t-1}} \exp\left(g_t\right) \left(\frac{i_t}{i_{t-1}} \exp\left(g_t\right) - exp\left(g_I\right) \right) \right]$$

Using the government's budget constraint we can solve for the lump sum taxes, $T_t = Z_t X_t$, which, along with the law of motion of net foreign assets and using the fact that real dividends received by the household are just the sum of real profits from intermediate good firms, we can replace in the budget constraint and obtain that

$$C_t + I_t + X_t + TB_t = Y_t,$$

i.e. the resource constraint of the economy. Its stationarized version is given by:

$$c_t + i_t + x_t + tb_t = y_t,$$

and the stationarized monopolistic dividends equation is:

$$d_{t} = y_{t} - r_{t}^{K} k_{t-1} \exp(-g_{t}) - w_{t} l_{t}.$$

We can also write the law of motion of net foreign assets as a stationarized equation:

$$b_t = (1 + r_{t-1})b_{t-1} \exp(-g_t) - d_t F_t + t b_t.$$

The real interest rate on foreign assets or liabilities is

$$r_t = \bar{r} + \psi \left[\exp \left(\frac{\bar{b}}{\bar{y}} - \frac{b_t}{y_t} \right) - 1 \right].$$

The current account in stationarized form is:

$$ca_{t} = tb_{t} - d_{t}F_{t} + r_{t-1}b_{t-1}\exp(-g_{t}).$$

The aggregate production function can be written in terms of stationarized variables as:

$$y_t = \exp\left(\epsilon_t^{\text{temp}}\right) \left(a_t\right)^{\alpha} \left(l_t\right)^{\alpha} \left[k_{t-1} \exp\left(-g_t\right)\right]^{1-\alpha}$$

Moreover, the demand equation for capital is:

$$r_t^k = \frac{\xi - 1}{\xi} (1 - \alpha) \frac{y_t}{k_{t-1}} \exp(g_t),$$

and for labor is:

$$w_t = \frac{\xi - 1}{\xi} \alpha \frac{y_t}{l_t}.$$

MNCs' entry probability function is stationarized as:

$$f_t = 1 - \exp\left(-\frac{m_t \left[1 - \eta_f \left(\frac{m_t}{m_{t-1}}\right)\right] \exp(\epsilon_t^f)}{\varpi^f}\right),$$

where

$$\eta_f\left(\frac{m_t}{m_{t-1}}\right) = \frac{\gamma_f}{2} \left(\frac{m_t}{m_{t-1}} \exp\left(g_t\right) - \exp\left(g_M\right)\right)^2.$$

The first order condition with respect to FDI is

$$p_t^{S\max} \exp\left(-\frac{m_t \left[1 - \eta_f \left(\frac{m_t}{m_{t-1}}\right)\right] \exp(\epsilon_t^f)}{\varpi^f}\right)$$
$$\frac{\left[1 - \eta_f \left(\frac{m_t}{m_{t-1}}\right) - \gamma_f \frac{m_t}{m_{t-1}} \left(\frac{m_t}{m_{t-1}} \exp\left(g_t\right) - \exp\left(g_M\right)\right)\right] \exp(\epsilon_t^f)}{\varpi^f} = R_{\rm M}.$$

Production function of innovation probability:

$$\frac{x_t \left[1 - \eta_X \left(\frac{x_t}{x_{t-1}}\right)\right]}{\varpi^{RD} a_t^{\text{target}}} = \ln\left(\frac{1}{1 - n_t}\right),$$

where

$$\eta_X\left(\frac{x_t}{x_{t-1}}\right) = \frac{\gamma_{RD}}{2} \left(\frac{x_t}{x_{t-1}} \exp\left(g_t\right) - \exp\left(g_X\right)\right)^2.$$

R&D first order conditions:

$$\frac{x_t \left(1 - \eta_X \left(\frac{x_t}{x_{t-1}}\right)\right)}{\varpi^{RD} a_t^{\text{target}}} = \ln\left(\frac{p_t^S \left(1 - f_t\right) \left[1 - \eta_X \left(\frac{x_t}{x_{t-1}}\right) - \gamma_{RD} \exp\left(g_t\right) \frac{x_t}{x_{t-1}} \left(\frac{x_t}{x_{t-1}} \exp\left(g_t\right) - exp\left(g_X\right)\right)\right]}{\varpi^{RD} a_t^{\text{target}} \left(1 - Z_t^{RD}\right) R_{\text{RD}}}\right).$$

The average manufacturing firm value p^{S} in stationarized form is obtained from:

$$\lambda_t p_t^S = \mathbb{E}_t \beta \lambda_{t+1} \exp((g_{t+1}))^{1-\sigma} \left[d_{t+1} + p_{t+1}^S (1 - n_{t+1}) (1 - f_{t+1}) \right].$$

Frontier productivity-adjusted average target technology:

$$a_t^{\text{target}} = a_{t-1} \exp(-g_t) + \phi_{t-1} \left[1 - a_{t-1} \exp(-g_t)\right],$$

with ϕ_t defined by:

$$\phi_t = \phi_0^{(1-a_t)}.$$

Law of motion of domestic technology:

$$a_{t} = f_{t-1} + (1 - f_{t-1}) \left[n_{t-1} a_{t}^{\text{target}} + (1 - n_{t-1}) a_{t-1} \exp\left(-g_{t}\right) \right].$$

Firm values are proportional to their productivity index:

$$p_t^{S\max} = \frac{p_t^S}{a_t^{\text{target}}}.$$

Since all variables involved are stationary, the dynamics of the set of sectors dominated by MNCs remains the same:

$$F_t = F_{t-1} + (1 - F_{t-1})f_{t-1} - F_{t-1}(1 - f_{t-1})n_{t-1}.$$

Similarly, the following equations also remain unchanged:

$$Z_t = Z \exp(\epsilon_t^{RD}),$$

$$\epsilon_t^{\text{temp}} = \rho_{\text{temp}} \epsilon_{t-1}^{\text{temp}} + u_t^{\text{temp}},$$

$$\epsilon_t^{\text{beta}} = \rho_{\text{beta}} \epsilon_{t-1}^{\text{beta}} + u_t^{\text{beta}},$$

$$\epsilon_t^{MEI} = \rho_{MEI} \epsilon_{t-1}^{MEI} + u_t^{MEI},$$

$$\epsilon_t^f = \rho_f \epsilon_{t-1}^f + u_t^f,$$

$$\epsilon_t^g = \rho_g \epsilon_{t-1}^g + u_t^g, \text{ and}$$

$$\epsilon_t^{RD} = \rho_{RD} \epsilon_{t-1}^{RD} + u_t^{RD}.$$

Therefore we have 28 independent equations for 28 endogenous variables: c_t, λ_t, r_t, l_t , $\begin{array}{l} w_{t}, y_{t}, a_{t}, a_{t}^{\mathrm{target}}, k_{t}, r_{t}^{K}, i_{t}, p_{t}^{K}, d_{t}, n_{t}, p_{t}^{S}, p_{t}^{S\mathrm{max}}, x_{t}, b_{t}, tb_{t}, ca_{t}, F_{t}, m_{t}, \phi_{t}, \pi_{t}, s_{t}, f_{t}, Z_{t}, g_{t}, \\ plus six shocks with corresponding autoregressive processes <math>\epsilon_{t}^{\mathrm{MEI}}, \epsilon_{t}^{\mathrm{beta}}, \epsilon_{t}^{t}, \epsilon_{t}^{g}, \epsilon_{t}^{g}$ and ϵ_{t}^{RD} . The parameters are: $\sigma, \chi, \xi, \alpha, g_{\mathrm{exo}}, \beta, h, \omega, \theta, \gamma_{K}, \delta, \nu, \gamma_{RD}, \varpi^{RD}, R_{RD}, \gamma_{f}, \varpi^{f}, R_{\mathrm{M}}, \end{array}$

 $\bar{r}, \bar{b}, \bar{y}, \bar{\psi}, Z$, and the autocorrelation coefficients $\rho_g, \rho_f, \rho_{temp}, \rho_{RD}, \rho_{beta}, \rho_{MEI}$.

C Estimation

C.1 Identification



Figure 12: Prior mean - Identification using info from observables.



Figure 13: Prior mean - Sensitivity plot.



Figure 14: Prior mean - Identification patterns (Information matrix): SMALLEST SV.



Figure 15: Prior mean - Identification patterns (Information matrix): HIGHEST SV.



Figure 16: Prior mean - Collinearity patterns with 1 parameter(s).



Figure 17: Prior mean - Collinearity patterns with 2 parameter(s).



C.2 Prior and Posterior Distribution

Figure 18: Prior distribution for the estimated parameters.



Figure 19: Posterior distribution for the estimated parameters.



Figure 20: Priors and posteriors .The grey line defines the prior density, while the black line represents the density of the posterior distribution. The green horizontal line marks the posterior mode.

C.3 Convergence Checks



Figure 21: Multivariate convergence diagnostics for the Metropolis-Hastings. The first, second and third rows are respectively the criteria based on the eighty percent interval, the second and third moments. The different parameters are aggregated using the posterior kernel.



Figure 22: Univariate convergence diagnostics for the Metropolis-Hastings (standard deviations). The first, second and third columns are respectively the criteria based on the eighty percent interval, the second and third moments.



Figure 23: Univariate convergence diagnostics for the Metropolis-Hastings (autocorrelation coefficients). The first, second and third columns are respectively the criteria based on the eighty percent interval, the second and third moments.



Figure 24: Univariate convergence diagnostics for the Metropolis-Hastings (model parameters). The first, second and third columns are respectively the criteria based on the eighty percent interval, the second and third moments.



Figure 25: Trace plot for the posterior density



Figure 26: Trace plot for the autocorrelation coefficients



Figure 27: Trace plot for the model parameters


Figure 28: Trace plot for the standard deviation of structural shocks

C.5 Summary of Parameters

Parameter	Description	Value
Calibrated Parameters		
α	Labor share	0.56
δ	Capital depreciation rate	0.01
β	Subjective discount factor	0.9875
$g_{\rm exo}$	Steady state frontier technology growth rate	0.0023
\bar{r}	Steady state rest of the world interest rate	0.015
n	Steady state success probability of domestic innovation	0.0208
C/Y	Steady state consumption as a share of GDP	0.53
l	Steady state hours worked	1/3
θ	Parameter determining the elasticity of labor supply	2
σ	Parameter determining the intertemporal elasticity of substitution	2
F_{\perp}	Steady state share of MNCs in the economy	0.5
X/Y	Steady state BERD as a share of GDP	0.006
Z	Steady state government subsidy as a share of BERD	0.115
FDI/Y	Steady state FDI as a share of GDP	0.0415
Estimated Parameters		
h	Internal consumption habit	0.855
χ	Parameter determining the size of the wealth effect	0.931
γ_K	Capital investment adjustment cost	6.165
γ_{RD}	R&D investment adjustment cost	4.414
γ_f	MNCs investment adjustment cost	0.025
ψ	Sensitivity parameter of the interest rate premium	0.003
ξ	Elasticity of substitution between intermediate goods	7.047
ν	Speed of productivity convergence	0.104
Autocorrelation Coefficients		
ρ^{g}	Shock to the frontier technology	0.275
$ ho^{ ext{temp}}$	Temporary productivity shock	0.989
$ ho^{ m beta}$	Shock to household's preferences	0.681
ρ^{MEI}	Marginal efficiency to investment shock	0.357
$ ho^{ m f}$	MNCs' entry rate shock	0.347
ρ^{RD}	Shock to R&D subsidy	0.798
Standard Deviations		
$SE_{u^{g}}$	Shock to the frontier technology	0.010
$SE_{u^{\text{temp}}}$	Temporary productivity shock	0.014
$SE_{u^{\text{beta}}}$	Shock to household's preferences	0.078
$SE_{u^{\text{MEI}}}$	Marginal efficiency to investment shock	0.201
$SE_{u^{\mathrm{f}}}$	MNCs' entry rate shock	0.318
$SE_{u^{\rm RD}}$	Shock to R&D subsidy	0.260
Endogenous Parameters		
ω	Labor disutility	5.890
ϖ^{RD}	R&D difficulty	1.241
$R_{\rm RD}$	R&D fixed cost	8.781
ϖ^f	MNCs' entry difficulty	8.536
$R_{\rm M}$	MNCs fixed entry cost	1.144

Table 7: Summary of Parameters

C.6 Other Estimation Results



Figure 29: Historical and smoothed variables.



Figure 30: Smoothed shocks.